FIRCREST CITY COUNCIL REGULAR MEETING AGENDA

TUESDAY, JULY 9, 2019 7:00 P.M.

COUNCIL CHAMBERS FIRCREST CITY HALL, 115 RAMSDELL STREET

- 1. CALL TO ORDER
- 2. PLEDGE OF ALLEGIANCE
- 3. ROLL CALL
- 4. PRESIDING OFFICER'S REPORT
 - A. Proclamation: National Night Out
 - **B.** Pool and Bathhouse Bid Award and Contract Discussion

5. PUBLIC COMMENTS FOR ITEMS NOT ON THE AGENDA

(Please sign the Public Attendance Roster/Public Hearing Sign-Up Sheet if you wish to speak during the meeting.)

6. COMMITTEE, COMMISSION & LIAISON REPORTS

- A. Administration
- B. Environment, Planning and Building
- **C.** Finance, IT, Facilities
- **D.** Other Liaison Reports

7. CONSENT CALENDAR

- A. Approval of vouchers/payroll checks
- **B.** Approval of minutes: Ju

June 17, 2019 Study Session

June 25, 2019 Regular Meeting July 1, 2019 Special Meeting

8. PUBLIC HEARING 7:15 P.M.

9. UNFINISHED BUSINESS

10. NEW BUSINESS

- A. Motion: Alameda Grind and Overlay Project Bid Award and Contract Approval
- B. Resolution: KPG, P.S. Supplement #1 for Construction Management Services

11. CITY MANAGER COMMENTS

- **12. DEPARTMENT HEAD COMMENTS**
- **13. COUNCILMEMBER COMMENTS**
- **14. EXECUTIVE SESSION**

15. ADJOURNMENT

COUNCIL MAY ADD AND TAKE ACTION ON OTHER ITEMS NOT LISTED ON AGENDA

PRESIDING OFFICER'S REPORT: Proclamation: National Night Out ITEM 4A.

FROM:

John Cheesman, Police Chief

RECOMMENDED MOTION: I move to authorize the Mayor's signature on a proclamation proclaiming August 6, 2019 as "National Night Out" in the City of Fircrest.

PROPOSAL: The Council is being asked to approve the National Night Out proclamation.

FISCAL IMPACT: None.

ADVANTAGE: Support of this program and cooperation with other cities and counties show solidarity of citizens standing against crime, illegal drugs, and violence. This sends out a strong message that this type of activity will not be tolerated in the City of Fircrest.

DISADVANTAGES: None known.

ALTERNATIVES: Not to participate.

HISTORY: The National Association of Town Watch (NATW) has asked the City to support the 36th annual National Night Out and to issue a proclamation in support of the program.

ATTACHMENT: Proclamation



WHEREAS, The National Association of Town Watch (NATW) is sponsoring a unique, nationwide crime, drug, and violence prevention program on August 6, 2019, called National Night Out; and

WHEREAS, the 36th Annual National Night Out provides a unique opportunity for the City of Fircrest to join forces with thousands of other communities across the country in promoting cooperative, police-community crime-fighting efforts; and

WHEREAS, Fircrest is unique in its ambience, closeness of neighbors, awareness of community, and cooperation and concern for each other; and

WHEREAS, the citizens of Fircrest play a vital role in assisting the Fircrest Police Department through joint crime and drug prevention programs; and

WHEREAS, it is essential that all citizens of Fircrest be aware of the importance of crime prevention programs and the impact their participation can have on reducing crime, illegal drugs, and violence in Fircrest; and

WHEREAS, the police-community partnerships and neighborhood safety and awareness and cooperation are important themes of the National Night Out Program.

NOW, THEREFORE, WE, MAYOR AND COUNCILMEMBERS, of the City of Fircrest, do hereby proclaim August 6, 2019, as

National Night Out in the City of Fircrest, Washington, and urge all citizens to join the Fircrest Police Department and the National Association of Town Watch in participating in this important event.

Passed this 9th day of July 2019

Denny Waltier, Mayor Pro Tempore

FIRCREST POOL AND BATHHOUSE PROJECT Division 00 – Procurement and Contracting Requirements May 30, 2019

OWNER/ CONTRACTOR CONTRACT

THIS AGREEMENT, made and entered into in duplicate, this _____day of _____, 20____ by and between City of Fircrest, hereinafter called the Owner, and hereinafter called the Contractor,

WITNESSETH:

That in consideration of the terms and conditions contained herein, and attached and made a part of this agreement, the parties hereto covenant and agree as follows:

I. The Contractor shall do all work and furnish all tools, materials, and equipment for <u>FIRCREST</u> <u>COMMUNITY POOL AND BATHHOUSE PROJECT</u> in accordance with and as described in the attached plans and specifications which are by this reference incorporated herein and made a part hereof, and shall perform any alterations in or additions to the work provided under this contract and every part thereof.

Work shall be completed within two hundred and eighty-eight (288) calendar days from Notice to Proceed. The project site will be available as early as August 19, 2019 and the project needs to have Substantial Completion by June 1, 2020 and Final Completion by June 12th, 2020.

If said work is not completed within the time specified, the Contractor agrees to pay to the Owner, as liquidated damages, the sum ONE THOUSAND FIVE HUNDRED DOLLARS (\$1,500) for each and every calendar day the work remains incomplete, exclusive of those days wherefore the Owner has granted an extension of time.

The Contractor hereby agrees the amount set forth above is a reasonable estimate of actual damage which would be caused by the failure to substantially complete the work on time.

The Contractor shall provide and bear the expense of all equipment, work and labor of any sort whatsoever that may be required for the transfer of materials and for constructing and completing the work provided for in this contract and every part thereof, except such as are mentioned in the specifications to be furnished by the City of Fircrest.

II. The City of Fircrest hereby promises and agrees with the Contractor to employ, and does employ the Contractor to provide the materials and to do and cause to be done the above described work and to complete and finish the same according to the attached plans and specifications and the terms and conditions herein contained; and hereby contracts to pay for the same according to the attached specifications for the itemized bid price of \$______,

at the time and in the manner and upon the conditions provided for in this contract.

The parties shall be bound by the Constitution and Laws of the State of Washington, and the Fircrest Municipal Code, and by all applicable federal laws and government regulations, which provisions are incorporated by reference herein.

For the convenience of the parties of this Contract it is mutually agreed that any claims or causes of action which the Contractor has against the Owner arising from this Contract shall be brought within 180 calendar days from the Completion Date of the Contract. It is further agreed by the parties that any such claims, disputes, or causes of action which cannot be resolved pursuant to the procedures set forth in the Contract Documents shall be brought only in the Superior Court of Pierce County. The parties understand and agree that the Contractor's failure to bring suit within the time period provided shall be a complete bar to any such claims or causes of action.

III. In accordance with RCW Chapter 39.12 and the Project Manual, the Contractor shall pay, or cause to be paid to persons employed on or in connection with the Fircrest Community Pool and Bathhouse Project, not less than the prevailing rate of wage for an hour's work specified for the labor performed.

IV. The Contractor for himself, and for his/her heirs, executors, administrators, successors, and assigns, does hereby agree to the full performance of all the covenants herein upon the part of the Contractor.

V. It is further provided that no liability shall attach to the City of Fircrest by reason of entering into this contract, except as expressly provided herein.

VI. The Contractor shall not discriminate against any person, firm, partnership, or organization as it pertains to race, color, religion, sex, age, national origin, marital status, sexual orientation, medical condition, physical handicap or disability. Any person, firm, partnership, or organization contracting with, or doing business with, the City shall be in conformity with the City's policy on non-discrimination.

VII. If the Contractor violates any material covenant or provision of this Contract the Owner may: withhold payment due on any work done under the Contract until the Contractor complies with the Contract; order that work be stopped, and/or terminate the Contract.

VIII. The Contractor shall be responsible for maintaining, during the term of this Agreement and at its sole cost and expense, the types of insurance coverage and in the amounts described below. The Contractor shall furnish evidence, satisfactory to the City, of all such policies. During the term hereof, the Contractor shall take out and maintain in full force and effect the following insurance policies:

- A. Comprehensive public liability insurance, including General Liability, insuring the City and the Contractor against loss or liability for damages for personal injury, death or property damage arising out of or in connection with the performance by the Contractor of its obligations hereunder, with minimum liability limits of \$5,000,000 combined single limit for personal injury, death or property damage in any one occurrence.
- B. Such workmen's compensation and other similar insurance as may be required by law.
- C. Professional liability insurance with minimum liability limits of \$1,000,000.

Any payment of deductible shall be the sole responsibility of the Contractor. City shall be named as a primary non-contributory additional insured on the General Liability policies described above. The insurance policies shall (1) state that coverage shall apply separately to each insured against whom claim is made or suit is brought, except with respect to the limits of the insurer's liability; (2) shall be primary insurance with

regard to City; and (3) shall state that the City will be given at least 30 days prior written notice of any cancellation, suspension or material change in coverage. The limits of liability required above are minimum required limits only and do not relieve the Contractor of claims that may exceed these minimum limits.

IX. With the exception of the sole negligence of the City, Consultant shall protect, defend, indemnify and hold harmless the City, its employees, directors, officers, and Councilmembers from any and all claims, demands, losses, actions, and liabilities to or by any person or entity arising out of or resulting from the acts or omissions of Consultant, its officers, and employees and its agents in performing this Agreement.

IN WITNESS WHEREOF the parties hereto have caused this agreement to be executed the day and year first hereinabove written.

By _____ (City Manager)

Ву _____

(Contractor)

Approved as to Legality:

(City Attorney)

$\mathbf{W}AIA^{\circ}$ Document A201["] – 2017

General Conditions of the Contract for Construction

for the following PROJECT:

Firerest Pool and Bathhouse 545 Contra Costa Street Fircrest, WA 98466

THE OWNER:

City of Firerest Parks and Recreation 555 Contra Costa Street Fircrest, WA 98466

THE ARCHITECT:

ARC Architects, Inc. 119 S. Main Street, #200 Seattle, WA 98104 206-322-3322

TABLE OF ARTICLES

- **GENERAL PROVISIONS** 1
- 2 OWNER
- 3 CONTRACTOR
- ARCHITECT
- 5 **SUBCONTRACTORS**
- CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS 6
- CHANGES IN THE WORK 7
- 8 TIME
- 9 **PAYMENTS AND COMPLETION**
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 INSURANCE AND BONDS
- UNCOVERING AND CORRECTION OF WORK 12
- 13 **MISCELLANEOUS PROVISIONS**
- **TERMINATION OR SUSPENSION OF THE CONTRACT** 14

15 CLAIMS AND DISPUTES

ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

For guidance in modifying this document to include supplementary conditions, see AIA Document A503[™], Guide for Supplementary Conditions.

AlA Document A201TM – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 00:42:00 ET on 05/30/2019 under Order Init 1 No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

ARTICLE 1 GENERAL PROVISIONS § 1.1 Basic Definitions

§ 1.1.1 The Contract Documents

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement, and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive, or (4) a written order for a minor change in the Work issued by the Architect. Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposal, or portions of Addenda relating to bidding or proposal requirements.

§ 1.1.2 The Contract

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, (3) between the Owner and the Architect or the Architect's consultants, or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties.

§ 1.1.3 The Work

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment, and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project.

§ 1.1.4 The Project

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by Separate Contractors.

§ 1.1.5 The Drawings

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules, and diagrams.

§ 1.1.6 The Specifications

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services.

§ 1.1.7 Instruments of Service

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

§ 1.1.8 Initial Decision Maker

The Initial Decision Maker is the person identified in the Agreement to render initial decisions on Claims in accordance with Section 15.2. The Initial Decision Maker shall not show partiality to the Owner or Contractor and shall not be liable for results of interpretations or decisions rendered in good faith.

§ 1.2 Correlation and Intent of the Contract Documents

§ 1.2.1 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

Init

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

§ 1.2.1.1 The invalidity of any provision of the Contract Documents shall not invalidate the Contract or its remaining provisions. If it is determined that any provision of the Contract Documents violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Contract Documents shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Contract.

§ 1.2.2 Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade.

§ 1.2.3 Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings.

§ 1.3 Capitalization

Terms capitalized in these General Conditions include those that are (1) specifically defined, (2) the titles of numbered articles, or (3) the titles of other documents published by the American Institute of Architects.

§ 1.4 Interpretation

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "thc" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

§ 1.5 Ownership and Use of Drawings, Specifications, and Other Instruments of Service

§ 1.5.1 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and retain all common law, statutory, and other reserved rights in their Instruments of Service, including copyrights. The Contractor, Subcontractors, Sub-subcontractors, and suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with the Project is not to be construed as publication in derogation of the Architect's or Architect's consultants' reserved rights.

§ 1.5.2 The Contractor, Subcontractors, Sub-subcontractors, and suppliers are authorized to use and reproduce the Instruments of Service provided to them, subject to any protocols established pursuant to Sections 1.7 and 1.8, solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and suppliers may not use the Instruments of Service on other projects or for additions to the Project outside the scope of the Work without the specific written consent of the Owner, Architect, and the Architect's consultants.

§ 1.6 Notice

§ 1.6.1 Except as otherwise provided in Section 1.6.2, where the Contract Documents require one party to notify or give notice to the other party, such notice shall be provided in writing to the designated representative of the party to whom the notice is addressed and shall be deemed to have been duly served if delivered in person, by mail, by courier, or by electronic transmission if a method for electronic transmission is set forth in the Agreement.

§ 1.6.2 Notice of Claims as provided in Section 15.1.3 shall be provided in writing and shall be deemed to have been duly served only if delivered to the designated representative of the party to whom the notice is addressed by certified or registered mail, or by courier providing proof of delivery.

§ 1.7 Digital Data Use and Transmission

The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. The parties will use AIA Document E203TM-2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.

§ 1.8 Building Information Models Use and Reliance

Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in AIA Document E203TM-2013, Building Information Modeling and Digital Data Exhibit, and the requisite AIA Document

(1848930370)

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be Init. 1 prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No 9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

G202TM–2013, Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

ARTICLE 2 OWNER

§ 2.1 General

§ 2.1.1 The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.

§ 2.1.2 The Owner shall furnish to the Contractor, within fifteen days after receipt of a written request, information necessary and relevant for the Contractor to evaluate, give notice of, or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein.

§ 2.2 Evidence of the Owner's Financial Arrangements

§ 2.2.1 Prior to commencement of the Work and upon written request by the Contractor, the Owner shall furnish to the Contractor reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. The Contractor shall have no obligation to commence the Work until the Owner provides such evidence. If commencement of the Work is delayed under this Section 2.2.1, the Contract Time shall be extended appropriately.

§ 2.2.2 Following commencement of the Work and upon written request by the Contractor, the Owner shall furnish to the Contractor reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract only if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due; or (3) a change in the Work materially changes the Contract Sum. If the Owner fails to provide such evidence, as required, within fourteen days of the Contractor's request, the Contractor may immediately stop the Work and, in that event, shall notify the Owner that the Work has stopped. However, if the request is made because a change in the Work materially changes the Contract Sum under (3) above, the Contractor may immediately stop only that portion of the Work affected by the change until reasonable evidence is provided. If the Work is stopped under this Section 2.2.2, the Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided in the Contract Documents.

§ 2.2.3 After the Owner furnishes evidence of financial arrangements under this Section 2.2, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.

§ 2.2.4 Where the Owner has designated information furnished under this Section 2.2 as "confidential," the Contractor shall keep the information confidential and shall not disclose it to any other person. However, the Contractor may disclose "confidential" information, after seven (7) days' notice to the Owner, where disclosure is required by law, including a subpoena or other form of compulsory legal process issued by a court or governmental entity, or by court or arbitrator(s) order. The Contractor may also disclose "confidential" information to its employees, consultants, surcties, Subcontractors and their employees, Sub-subcontractors, and others who need to know the content of such information solely and exclusively for the Project and who agree to maintain the confidentiality of such information.

§ 2.3 Information and Services Required of the Owner

§ 2.3.1 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities.

§ 2.3.2 The Owner shall retain an architect lawfully licensed to practice architecture, or an entity lawfully practicing architecture, in the jurisdiction where the Project is located. That person or entity is identified as the Architect in the Agreement and is referred to throughout the Contract Documents as if singular in number.

Init. 1

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treatles. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No 9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 2.3.3 If the employment of the Architect terminates, the Owner shall employ a successor to whom the Contractor has no reasonable objection and whose status under the Contract Documents shall be that of the Architect.

§ 2.3.4 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work.

§ 2.3.5 The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.

§ 2.3.6 Unless otherwise provided in the Contract Documents, the Owner shall furnish to the Contractor one copy of the Contract Documents for purposes of making reproductions pursuant to Section 1.5.2.

§ 2.4 Owner's Right to Stop the Work

If the Contractor fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 or repeatedly fails to carry out Work in accordance with the Contract Documents, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3.

§ 2.5 Owner's Right to Carry Out the Work

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a ten-day period after receipt of notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such default or neglect. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect and the Architect may, pursuant to Section 9.5.1, withhold or nullify a Certificate for Payment in whole or in part, to the extent reasonably necessary to reimburse the Owner for the reasonable cost of correcting such deficiencies, including Owner's expenses and compensation for the Architect's additional services made necessary by such default, neglect, or failure. If current and future payments are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner. If the Contractor disagrees with the actions of the Owner or the Architect, or the amounts claimed as costs to the Owner, the Contractor may file a Claim pursuant to Article 15.

ARTICLE 3 CONTRACTOR

§ 3.1 General

§ 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor shall designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.

§ 3.1.2 The Contractor shall perform the Work in accordance with the Contract Documents.

§ 3.1.3 The Contractor shall not be relieved of its obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor.

§ 3.2 Review of Contract Documents and Field Conditions by Contractor

§ 3.2.1 Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed, and correlated personal observations with requirements of the Contract Documents.

Init. 1

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

§ 3.2.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.3.4, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Contractor shall promptly report to the Architect any errors, inconsistencies or omissions discovered by or made known to the Contractor as a request for information in such form as the Architect may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.

§ 3.2.3 The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require.

§ 3.2.4 If the Contractor believes that additional cost or time is involved because of clarifications or instructions the Architect issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2 or 3.2.3, the Contractor shall submit Claims as provided in Article 15. If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner, subject to Section 15.1.7, as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities.

§ 3.3 Supervision and Construction Procedures

§ 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences, and procedures, and for coordinating all portions of the Work under the Contract. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences, or procedures, the Contractor shall evaluate the jobsite safety thereof and shall be solely responsible for the jobsite safety of such means, methods, techniques, sequences, or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely notice to the Owner and Architect, and shall propose alternative means, methods, techniques, sequences, or procedures. The Architect shall evaluate the proposed alternative solely for conformance with the design intent for the completed construction. Unless the Architect objects to the Contractor's proposed alternative, the Contractor shall perform the Work using its alternative means, methods, techniques, sequences, or procedures.

§ 3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of, the Contractor or any of its Subcontractors.

§ 3.3.3 The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

§ 3.4 Labor and Materials

§ 3.4.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.

§ 3.4.2 Except in the case of minor changes in the Work approved by the Architect in accordance with Section 3.12.8 or ordered by the Architect in accordance with Section 7.4, the Contractor may make substitutions only with the consent of the Owner, after evaluation by the Architect and in accordance with a Change Order or Construction Change Directive.

Init. 1

AIA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA* Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

§ 3.5 Warranty

§ 3.5.1 The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

§ 3.5.2 All material, equipment, or other special warranties required by the Contract Documents shall be issued in the name of the Owner, or shall be transferable to the Owner, and shall commence in accordance with Section 9.8.4.

§ 3.6 Taxes

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect.

§ 3.7 Permits, Fees, Notices and Compliance with Laws

§ 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit as well as for other permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations concluded.

§ 3.7.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work.

§ 3.7.3 If the Contractor performs Work knowing it to be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.

§ 3.7.4 Concealed or Unknown Conditions

If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 14 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend that an equitable adjustment be made in the Contract Sum or Contract Time, or both. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor, stating the reasons. If either party disputes the Architect's determination or recommendation, that party may submit a Claim as provided in Article 15.

§ 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 15.

Init. 1

AlA Document A201™ - 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

§ 3.8 Allowances

§ 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents, Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection.

§ 3.8.2 Unless otherwise provided in the Contract Documents,

- allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and all .1 required taxes, less applicable trade discounts;
- .2 Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit, and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowances; and
- .3 whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2.

§ 3.8.3 Materials and equipment under an allowance shall be selected by the Owner with reasonable promptness.

§ 3.9 Superintendent

§ 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor.

§ 3.9.2 The Contractor, as soon as practicable after award of the Contract, shall notify the Owner and Architect of the name and qualifications of a proposed superintendent. Within 14 days of receipt of the information, the Architect may notify the Contractor, stating whether the Owner or the Architect (1) has reasonable objection to the proposed superintendent or (2) requires additional time for review. Failure of the Architect to provide notice within the 14-day period shall constitute notice of no reasonable objection.

§ 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, which shall not unreasonably be withheld or delayed.

§ 3.10 Contractor's Construction and Submittal Schedules

§ 3.10.1 The Contractor, promptly after being awarded the Contract, shall submit for the Owner's and Architect's information a Contractor's construction schedule for the Work. The schedule shall contain detail appropriate for the Project, including (1) the date of commencement of the Work, interim schedule milestone dates, and the date of Substantial Completion; (2) an apportionment of the Work by construction activity; and (3) the time required for completion of each portion of the Work. The schedule shall provide for the orderly progression of the Work to completion and shall not exceed time limits current under the Contract Documents. The schedule shall be revised at appropriate intervals as required by the conditions of the Work and Project.

§ 3.10.2 The Contractor, promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, shall submit a submittal schedule for the Architect's approval. The Architect's approval shall not be unreasonably delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Architect reasonable time to review submittals. If the Contractor fails to submit a submittal schedule, or fails to provide submittals in accordance with the approved submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.

§ 3.10.3 The Contractor shall perform the Work in general accordance with the most recent schedules submitted to the Owner and Architect.

§ 3.11 Documents and Samples at the Site

1

The Contractor shall make available, at the Project site, the Contract Documents, including Change Orders, Construction Change Directives, and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and the approved Shop Drawings, Product Data, Samples, and similar required submittals. These shall be in electronic form or paper copy, available to the Architect and Owner, and

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1956, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Init. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed.

§ 3.12 Shop Drawings, Product Data and Samples

§ 3.12.1 Shop Drawings are drawings, diagrams, schedules, and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier, or distributor to illustrate some portion of the Work.

§ 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams, and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.

§ 3.12.3 Samples are physical examples that illustrate materials, equipment, or workmanship, and establish standards by which the Work will be judged.

§ 3.12.4 Shop Drawings, Product Data, Samples, and similar submittals are not Contract Documents. Their purpose is to demonstrate how the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review by the Architect is subject to the limitations of Section 4.2.7. Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Architect without action.

§ 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve, and submit to the Architect, Shop Drawings, Product Data, Samples, and similar submittals required by the Contract Documents, in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of Separate Contractors.

§ 3.12.6 By submitting Shop Drawings, Product Data, Samples, and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so, and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.

§ 3.12.7 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples, or similar submittals, until the respective submittal has been approved by the Architect.

§ 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from the requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples, or similar submittals, unless the Contractor has specifically notified the Architect of such deviation at the time of submittal and (1) the Architect has given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples, or similar submittals, by the Architect's approval thereof.

§ 3.12.9 The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples, or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such notice, the Architect's approval of a resubmission shall not apply to such revisions.

§ 3.12.10 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents for a portion of the Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences, and procedures. The Contractor shall not be required to provide professional services in violation of applicable law.

§ 3.12.10.1 If professional design services or certifications by a design professional related to systems, materials, or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Architect will specify all performance and design criteria that such services must satisfy. The Contractor shall be entitled to rely

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treatles. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

16

Init.

upon the adequacy and accuracy of the performance and design criteria provided in the Contract Documents. The Contractor shall cause such services or certifications to be provided by an appropriately licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings, and other submittals prepared by such professional. Shop Drawings, and other submittals related to the Work, designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Architect. The Owner and the Architect shall be entitled to rely upon the adequacy and accuracy of the services, certifications, and approvals performed or provided by such design professionals, provided the Owner and Architect have specified to the Contractor the performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.10, the Architect will review and approve or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents.

§ 3.12.10.2 If the Contract Documents require the Contractor's design professional to certify that the Work has been performed in accordance with the design criteria, the Contractor shall furnish such certifications to the Architect at the time and in the form specified by the Architect.

§ 3.13 Use of Site

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, lawful orders of public authoritics, and the Contract Documents and shall not unreasonably encumber the site with materials or equipment.

§ 3.14 Cutting and Patching

§ 3.14.1 The Contractor shall be responsible for cutting, fitting, or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting, or patching shall be restored to the condition existing prior to the cutting, fitting, or patching, unless otherwise required by the Contract Documents.

§ 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or Separate Contractors by cutting, patching, or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter construction by the Owner or a Separate Contractor except with written consent of the Owner and of the Separate Contractor. Consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold, from the Owner or a Separate Contractor, its consent to cutting or otherwise altering the Work.

§ 3.15 Cleaning Up

§ 3.15.1 The Contractor shall kccp the premises and surrounding area free from accumulation of waste materials and rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery, and surplus materials from and about the Project.

§ 3.15.2 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and the Owner shall be entitled to reimbursement from the Contractor.

§ 3.16 Access to Work

The Contractor shall provide the Owner and Architect with access to the Work in preparation and progress wherever located.

§ 3.17 Royalties, Patents and Copyrights

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for defense or loss when a particular design, process, or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications, or other documents prepared by the Owner or Architect. However, if an infringement of a copyright or patent is discovered by, or made known to, the Contractor, the Contractor shall be responsible for the loss unless the information is promptly furnished to the Architect.

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document Is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 3.18 Indemnification

§ 3.18.1 To the fullest extent permitted by law, the Contractor shall indemnify and hold harmless the Owner, Architect, Architect's consultants, and agents and employees of any of them from and against claims, damages, losses, and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work, provided that such claim, damage, loss, or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), but only to the extent caused by the negligent acts or omissions of the Contractor, a Subcontractor, anyone directly or indirectly employed by them, or anyone for whose acts they may be liable, regardless of whether or not such claim, damage, loss, or expense is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a party or person described in this Section 3.18.

§ 3.18.2 In claims against any person or entity indemnified under this Section 3.18 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them, or anyone for whose acts they may be liable, the indemnification obligation under Section 3.18.1 shall not be limited by a limitation on amount or type of damages, compensation, or benefits payable by or for the Contractor or a Subcontractor under workers' compensation acts, disability benefit acts, or other employee benefit acts.

ARTICLE 4 ARCHITECT

§ 4.1 General

§ 4.1.1 The Architect is the person or entity retained by the Owner pursuant to Section 2.3.2 and identified as such in the Agreement.

§ 4.1.2 Duties, responsibilities, and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified, or extended without written consent of the Owner, Contractor, and Architect. Consent shall not be unreasonably withheld.

§ 4.2 Administration of the Contract

§ 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Architect issues the final Certificate for Payment. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.

§ 4.2.2 The Architect will visit the site at intervals appropriate to the stage of construction, or as otherwise agreed with the Owner, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine in general if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Architect will not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents.

§ 4.2.3 On the basis of the site visits, the Architect will keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work. The Architect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect will not have control over or charge of, and will not be responsible for acts or omissions of, the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.

§ 4.2.4 Communications

The Owner and Contractor shall include the Architect in all communications that relate to or affect the Architect's services or professional responsibilities. The Owner shall promptly notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and suppliers shall be through the Contractor. Communications by and with Separate Contractors shall be through the Owner. The Contract Documents may specify other communication protocols.

Init. 1

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 4.2.5 Based on the Architect's evaluations of the Contractor's Applications for Payment, the Architect will review and certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.

§ 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect will have authority to require inspection or testing of the Work in accordance with Sections 13.4.2 and 13.4.3, whether or not the Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, suppliers, their agents or employees, or other persons or entities performing portions of the Work.

§ 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data, and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5, and 3.12. The Architect's review shall not constitute approval of safety precautions or of any construction means, methods, techniques, sequences, or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

§ 4.2.8 The Architect will prepare Change Orders and Construction Change Directives, and may order minor changes in the Work as provided in Section 7.4. The Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4.

§ 4.2.9 The Architect will conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9.8; receive and forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract and assembled by the Contractor pursuant to Section 9.10; and issue a final Certificate for Payment pursuant to Section 9.10.

§ 4.2.10 If the Owner and Architect agree, the Architect will provide one or more Project representatives to assist in carrying out the Architect's responsibilities at the site. The Owner shall notify the Contractor of any change in the duties, responsibilities and limitations of authority of the Project representatives.

§ 4.2.11 The Architect will interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness.

§ 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either, and will not be liable for results of interpretations or decisions rendered in good faith.

§ 4.2.13 The Architect's decisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.

§ 4.2.14 The Architect will review and respond to requests for information about the Contract Documents. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information.

Init.

AlA Document A201[™] – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

ARTICLE 5 SUBCONTRACTORS

§ 5.1 Definitions

§ 5.1.1 A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a Separate Contractor or the subcontractors of a Separate Contractor.

§ 5.1.2 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor.

§ 5.2 Award of Subcontracts and Other Contracts for Portions of the Work

§ 5.2.1 Unless otherwise stated in the Contract Documents, the Contractor, as soon as practicable after award of the Contract, shall notify the Owner and Architect of the persons or entities proposed for each principal portion of the Work, including those who are to furnish materials or equipment fabricated to a special design. Within 14 days of receipt of the information, the Architect may notify the Contractor whether the Owner or the Architect (1) has reasonable objection to any such proposed person or entity or (2) requires additional time for review. Failure of the Architect to provide notice within the 14-day period shall constitute notice of no reasonable objection.

§ 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.

§ 5.2.3 If the Owner or Architect has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Architect has no reasonable objection. If the proposed but rejected Subcontractor was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute Subcontractor's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Contractor has acted promptly and responsively in submitting names as required.

§ 5.2.4 The Contractor shall not substitute a Subcontractor, person, or entity for one previously selected if the Owner or Architect makes reasonable objection to such substitution.

§ 5.3 Subcontractual Relations

By appropriate written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work that the Contractor, by these Contract Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies, and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

§ 5.4 Contingent Assignment of Subcontracts

Init.

1

§ 5.4.1 Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that

- assignment is effective only after termination of the Contract by the Owner for cause pursuant to .1 Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor; and
- assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the .2 Contract.

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale, User Notes:

(1848930370)

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract.

§ 5.4.2 Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be equitably adjusted for increases in cost resulting from the suspension.

§ 5.4.3 Upon assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity. If the Owner assigns the subcontract to a successor contractor or other entity, the Owner shall nevertheless remain legally responsible for all of the successor contractor's obligations under the subcontract.

ARTICLE 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS § 6.1 Owner's Right to Perform Construction and to Award Separate Contracts

§ 6.1.1 The term "Separate Contractor(s)" shall mean other contractors retained by the Owner under separate agreements. The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and with Separate Contractors retained under Conditions of the Contract substantially similar to those of this Contract, including those provisions of the Conditions of the Contract related to insurance and waiver of subrogation.

§ 6.1.2 When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.

§ 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces and of each Separate Contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with any Separate Contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to its construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, Separate Contractors, and the Owner until subsequently revised.

§ 6.1.4 Unless otherwise provided in the Contract Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces or with Separate Contractors, the Owner or its Separate Contractors shall have the same obligations and rights that the Contractor has under the Conditions of the Contract, including, without excluding others, those stated in Article 3, this Article 6, and Articles 10, 11, and 12.

§ 6.2 Mutual Responsibility

§ 6.2.1 The Contractor shall afford the Owner and Separate Contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.

§ 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a Separate Contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly notify the Architect of apparent discrepancies or defects in the construction or operations by the Owner or Separate Contractor that would render it unsuitable for proper execution and results of the Contractor's Work. Failure of the Contractor to notify the Architect of apparent discrepancies or defects prior to proceeding with the Work shall constitute an acknowledgment that the Owner's or Separate Contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work. The Contractor shall not be responsible for discrepancies or defects in the construction or operations by the Owner or Separate Contractor that are not apparent.

§ 6.2.3 The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a Separate Contractor because of the Contractor's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Contractor for costs the Contractor incurs because of a Separate Contractor's delays, improperly timed activities, damage to the Work or defective construction.

§ 6.2.4 The Contractor shall promptly remedy damage that the Contractor wrongfully causes to completed or partially completed construction or to property of the Owner or Separate Contractor as provided in Section 10.2.5.

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Init. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on D5/30/2019 under Order 1 No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 6.2.5 The Owner and each Separate Contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Section 3.14.

§ 6.3 Owner's Right to Clean Up

If a dispute arises among the Contractor, Separate Contractors, and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and the Architect will allocate the cost among those responsible.

ARTICLE 7 CHANGES IN THE WORK

§ 7.1 General

§ 7.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.

§ 7.1.2 A Change Order shall be based upon agreement among the Owner, Contractor, and Architect. A Construction Change Directive requires agreement by the Owner and Architect and may or may not be agreed to by the Contractor. An order for a minor change in the Work may be issued by the Architect alone.

§ 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents. The Contractor shall proceed promptly with changes in the Work, unless otherwise provided in the Change Order, Construction Change Directive, or order for a minor change in the Work.

§ 7.2 Change Orders

§ 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Owner, Contractor, and Architect stating their agreement upon all of the following:

- .1 The change in the Work;
- .2 The amount of the adjustment, if any, in the Contract Sum; and
- .3 The extent of the adjustment, if any, in the Contract Time.

§ 7.3 Construction Change Directives

§ 7.3.1 A Construction Change Directive is a written order prepared by the Architect and signed by the Owner and Architect, directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions, the Contract Sum and Contract Time being adjusted accordingly.

§ 7.3.2 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order.

§ 7.3.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:

- Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to .1 permit evaluation;
- .2 Unit prices stated in the Contract Documents or subsequently agreed upon;
- Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or .3 percentage fee; or
- As provided in Section 7.3.4. .4

§ 7.3.4 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect shall determine the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.4 shall be limited to the following:

Init.

(1848930370)

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

- .1 Costs of labor, including applicable payroll taxes, fringe benefits required by agreement or custom, workers' compensation insurance, and other employee costs approved by the Architect;
- .2 Costs of materials, supplies, and equipment, including cost of transportation, whether incorporated or consumed;
- Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor .3 or others;
- .4 Costs of premiums for all bonds and insurance, permit fees, and sales, use, or similar taxes, directly related to the change; and
- .5 Costs of supervision and field office personnel directly attributable to the change.

§ 7.3.5 If the Contractor disagrees with the adjustment in the Contract Time, the Contractor may make a Claim in accordance with applicable provisions of Article 15.

§ 7.3.6 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time.

§ 7.3.7 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

§ 7.3.8 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

§ 7.3.9 Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in Applications for Payment. The Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the Architect's professional judgment, to be reasonably justified. The Architect's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.

§ 7.3.10 When the Owner and Contractor agree with a determination made by the Architect concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Architect will prepare a Change Order. Change Orders may be issued for all or any part of a Construction Change Directive.

§ 7.4 Minor Changes in the Work

The Architect may order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. The Architect's order for minor changes shall be in writing. If the Contractor believes that the proposed minor change in the Work will affect the Contract Sum or Contract Time, the Contractor shall notify the Architect and shall not proceed to implement the change in the Work. If the Contractor performs the Work set forth in the Architect's order for a minor change without prior notice to the Architect that such change will affect the Contract Sum or Contract Time, the Contractor waives any adjustment to the Contract Sum or extension of the Contract Time.

ARTICLE 8 TIME

§ 8.1 Definitions

§ 8.1.1 Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work,

§ 8.1.2 The date of commencement of the Work is the date established in the Agreement.

§ 8.1.3 The date of Substantial Completion is the date certified by the Architect in accordance with Section 9.8.

Init. 1

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 8.1.4 The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

§ 8.2 Progress and Completion

§ 8.2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement, the Contractor confirms that the Contract Time is a reasonable period for performing the Work.

§ 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, commence the Work prior to the effective date of insurance required to be furnished by the Contractor and Owner.

§ 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

§ 8.3 Delays and Extensions of Time

§ 8.3.1 If the Contractor is delayed at any time in the commencement or progress of the Work by (1) an act or neglect of the Owner or Architect, of an employee of either, or of a Separate Contractor; (2) by changes ordered in the Work; (3) by labor disputes, fire, unusual delay in deliveries, unavoidable casualties, adverse weather conditions documented in accordance with Section 15.1.6.2, or other causes beyond the Contractor's control; (4) by delay authorized by the Owner pending mediation and binding dispute resolution; or (5) by other causes that the Contractor asserts, and the Architect determines, justify delay, then the Contract Time shall be extended for such reasonable time as the Architect may determine.

§ 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Article 15.

§ 8.3.3 This Section 8.3 does not preclude recovery of damages for delay by either party under other provisions of the Contract Documents.

ARTICLE 9 PAYMENTS AND COMPLETION

§ 9.1 Contract Sum

§ 9.1.1 The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

§ 9.1.2 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed so that application of such unit prices to the actual quantities causes substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.

§ 9.2 Schedule of Values

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit a schedule of values to the Architect before the first Application for Payment, allocating the entire Contract Sum to the various portions of the Work. The schedule of values shall be prepared in the form, and supported by the data to substantiate its accuracy, required by the Architect. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment, Any changes to the schedule of values shall be submitted to the Architect and supported by such data to substantiate its accuracy as the Architect may require, and unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's subsequent Applications for Payment.

§ 9.3 Applications for Payment

1

§ 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2, for completed portions of the Work. The application shall be notarized, if required, and supported by all data substantiating the Contractor's right to payment that the Owner or Architect require, such as copies of requisitions, and releases and waivers of liens from Subcontractors and suppliers, and shall reflect retainage if provided for in the Contract Documents.

§ 9.3.1.1 As provided in Section 7.3.9, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Architect, but not yet included in Change Orders.

AIA Document A201TM – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Init. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

§ 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or supplier, unless such Work has been performed by others whom the Contractor intends to pay.

§ 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage, and transportation to the site, for such materials and equipment stored off the site.

§ 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information, and belief, be free and clear of liens, claims, security interests, or encumbrances, in favor of the Contractor, Subcontractors, suppliers, or other persons or entities that provided labor, materials, and equipment relating to the Work.

§ 9.4 Certificates for Payment

§ 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either (1) issue to the Owner a Certificate for Payment in the full amount of the Application for Payment, with a copy to the Contractor; or (2) issue to the Owner a Certificate for Payment for such amount as the Architect determines is properly due, and notify the Contractor and Owner of the Architect's reasons for withholding certification in part as provided in Section 9.5.1; or (3) withhold certification of the entire Application for Payment, and notify the Contractor and Owner of the Architect's reason for withholding certification in whole as provided in Section 9.5.1.

§ 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's evaluation of the Work and the data in the Application for Payment, that, to the best of the Architect's knowledge, information, and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion, and to specific qualifications expressed by the Architect. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work; (2) reviewed construction means, methods, techniques, sequences, or procedures; (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment; or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

§ 9.5 Decisions to Withhold Certification

§ 9.5.1 The Architect may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of

- .1 defective Work not remedied;
- .2 third party claims filed or reasonable evidence indicating probable filing of such claims, unless security acceptable to the Owner is provided by the Contractor;
- .3 failure of the Contractor to make payments properly to Subcontractors or suppliers for labor, materials or equipment;

AIA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA[®] Document is protected by U.S. Copyright Law and International Treaties. Inlt. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

- reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum; .4
- .5 damage to the Owner or a Separate Contractor;
- .6 reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
- .7 repeated failure to carry out the Work in accordance with the Contract Documents.

§ 9.5.2 When either party disputes the Architect's decision regarding a Certificate for Payment under Section 9.5.1, in whole or in part, that party may submit a Claim in accordance with Article 15.

§ 9.5.3 When the reasons for withholding certification are removed, certification will be made for amounts previously withheld.

§ 9.5.4 If the Architect withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or supplier to whom the Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Contractor shall reflect such payment on its next Application for Payment.

§ 9.6 Progress Payments

§ 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Architect.

§ 9.6.2 The Contractor shall pay each Subcontractor, no later than seven days after receipt of payment from the Owner, the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.

§ 9.6.3 The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor.

§ 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors and suppliers to ascertain whether they have been properly paid. Neither the Owner nor Architect shall have an obligation to pay, or to see to the payment of money to, a Subcontractor or supplier, except as may otherwise be required by law.

§ 9.6.5 The Contractor's payments to suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.

§ 9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.

§ 9.6.7 Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors or provided by suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, create any fiduciary liability or tort liability on the part of the Contractor for breach of trust, or entitle any person or entity to an award of punitive damages against the Contractor for breach of the requirements of this provision.

§ 9.6.8 Provided the Owner has fulfilled its payment obligations under the Contract Documents, the Contractor shall defend and indemnify the Owner from all loss, liability, damage or expense, including reasonable attorney's fccs and litigation expenses, arising out of any lien claim or other claim for payment by any Subcontractor or supplier of any tier. Upon receipt of notice of a lien claim or other claim for payment, the Owner shall notify the Contractor. If approved by the applicable court, when required, the Contractor may substitute a surety bond for the property against which the lien or other claim for payment has been asserted.

(1848930370)

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Init. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be 1 prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

§ 9.7 Failure of Payment

If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner does not pay the Contractor within seven days after the date established in the Contract Documents, the amount certified by the Architect or awarded by binding dispute resolution, then the Contractor may, upon seven additional days' notice to the Owner and Architect, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided for in the Contract Documents.

§ 9.8 Substantial Completion

§ 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use.

§ 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list docs not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.

§ 9.8.3 Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Architect to determine Substantial Completion.

§ 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion; establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance; and fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.

§ 9.8.5 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in the Certificate. Upon such acceptance, and consent of surety if any, the Owner shall make payment of retainage applying to the Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

§ 9.9 Partial Occupancy or Use

§ 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect.

§ 9.9.2 Immediately prior to such partial occupancy or use, the Owner, Contractor, and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

Init.

AIA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

§ 9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

§ 9.10 Final Completion and Final Payment

§ 9.10.1 Upon receipt of the Contractor's notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection. When the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect, (3) a written statement that the Contractor knows of no reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) consent of surety, if any, to final payment, (5) documentation of any special warranties, such as manufacturers' warranties or specific Subcontractor warranties, and (6) if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts and releases and waivers of liens, claims, security interests, or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien, claim, security interest, or encumbrance. If a lien, claim, security interest, or encumbrance remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging the lien, claim, security interest, or encumbrance, including all costs and reasonable attorneys' fees.

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed, corrected, and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of the surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of Claims.

§ 9.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising from

- liens, Claims, security interests, or encumbrances arising out of the Contract and unsettled; .1
- .2 failure of the Work to comply with the requirements of the Contract Documents;
- .3 terms of special warranties required by the Contract Documents; or
- .4 audits performed by the Owner, if permitted by the Contract Documents, after final payment.

§ 9.10.5 Acceptance of final payment by the Contractor, a Subcontractor, or a supplier, shall constitute a waiver of claims by that payce except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.

ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY

§ 10.1 Safety Precautions and Programs

The Contractor shall be responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the performance of the Contract.

§ 10.2 Safety of Persons and Property

§ 10.2.1 The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury, or loss to

AIA Document A201 M - 2017. Copyright @ 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The Init. American Institute of Architects, All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treatie Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be 1 prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

- 1 employees on the Work and other persons who may be affected thereby;
- .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody, or control of the Contractor, a Subcontractor, or a Sub-subcontractor; and
- .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures, and utilities not designated for removal, relocation, or replacement in the course of construction.

§ 10.2.2 The Contractor shall comply with, and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, bearing on safety of persons or property or their protection from damage, injury, or loss.

§ 10.2.3 The Contractor shall implement, erect, and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards; promulgating safety regulations; and notifying the owners and users of adjacent sites and utilities of the safeguards.

§ 10.2.4 When use or storage of explosives or other hazardous materials or equipment, or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.

§ 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3 caused in whole or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3. The Contractor may make a Claim for the cost to remedy the damage or loss to the extent such damage or loss is attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.

§ 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect.

§ 10.2.7 The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

§ 10.2.8 Injury or Damage to Person or Property

If either party suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, notice of the injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

§ 10.3 Hazardous Materials and Substances

§ 10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials or substances. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foresceable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and notify the Owner and Architect of the condition.

§ 10.3.2 Upon receipt of the Contractor's notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of the material or substance or who are to perform the task of removal or safe containment of the material or substance. The Contractor and the Architect will

(1848930370)

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Init. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order 1 No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or Architect has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor and the Architect have no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable additional costs of shutdown, delay, and start-up.

§ 10.3.3 To the fullest extent permitted by law, the Owner shall indemnify and hold harmless the Contractor, Subcontractors, Architect, Architect's consultants, and agents and employees of any of them from and against claims, damages, losses, and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, provided that such claim, damage, loss, or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), except to the extent that such damage, loss, or expense is due to the fault or negligence of the party seeking indemnity.

§ 10.3.4 The Owner shall not be responsible under this Section 10.3 for hazardous materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents. The Owner shall be responsible for hazardous materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or negligence in the use and handling of such materials or substances.

§ 10.3.5 The Contractor shall reimburse the Owner for the cost and expense the Owner incurs (1) for remediation of hazardous materials or substances the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.

§ 10.3.6 If, without negligence on the part of the Contractor, the Contractor is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall reimburse the Contractor for all cost and expense thereby incurred.

§ 10.4 Emergencies

1

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury, or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7.

ARTICLE 11 INSURANCE AND BONDS

§ 11.1 Contractor's Insurance and Bonds

§ 11.1.1 The Contractor shall purchase and maintain insurance of the types and limits of liability, containing the endorsements, and subject to the terms and conditions, as described in the Agreement or elsewhere in the Contract Documents. The Contractor shall purchase and maintain the required insurance from an insurance company or insurance companies lawfully authorized to issue insurance in the jurisdiction where the Project is located. The Owner, Architect, and Architect's consultants shall be named as additional insureds under the Contractor's commercial general liability policy or as otherwise described in the Contract Documents.

§ 11.1.2 The Contractor shall provide surety bonds of the types, for such penal sums, and subject to such terms and conditions as required by the Contract Documents. The Contractor shall purchase and maintain the required bonds from a company or companies lawfully authorized to issue surety bonds in the jurisdiction where the Project is located.

§ 11.1.3 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.

§ 11.1.4 Notice of Cancellation or Expiration of Contractor's Required Insurance. Within three (3) business days of the date the Contractor becomes aware of an impending or actual cancellation or expiration of any insurance required by the Contract Documents, the Contractor shall provide notice to the Owner of such impending or actual cancellation or

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA[®] Document is protected by U.S. Copyright Law and International Treaties. Init. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

expiration. Upon receipt of notice from the Contractor, the Owner shall, unless the lapse in coverage arises from an act or omission of the Owner, have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by the Contractor. The furnishing of notice by the Contractor shall not relieve the Contractor of any contractual obligation to provide any required coverage.

§ 11.2 Owner's Insurance

§ 11.2.1 The Owner shall purchase and maintain insurance of the types and limits of liability, containing the endorsements, and subject to the terms and conditions, as described in the Agreement or elsewhere in the Contract Documents. The Owner shall purchase and maintain the required insurance from an insurance company or insurance companies lawfully authorized to issue insurance in the jurisdiction where the Project is located.

§ 11.2.2 Failure to Purchase Required Property Insurance. If the Owner fails to purchase and maintain the required property insurance, with all of the coverages and in the amounts described in the Agreement or elsewhere in the Contract Documents, the Owner shall inform the Contractor in writing prior to commencement of the Work. Upon receipt of notice from the Owner, the Contractor may delay commencement of the Work and may obtain insurance that will protect the interests of the Contractor, Subcontractors, and Sub-Subcontractors in the Work. When the failure to provide coverage has been cured or resolved, the Contract Sum and Contract Time shall be equitably adjusted. In the event the Owner fails to procure coverage, the Owner waives all rights against the Contractor, Subcontractors, and Sub-subcontractors to the extent the loss to the Owner would have been covered by the insurance to have been procured by the Owner. The cost of the insurance shall be charged to the Owner by a Change Order. If the Owner does not provide written notice, and the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain the required insurance, the Owner shall reimburse the Contractor for all reasonable costs and damages attributable thereto.

§ 11.2.3 Notice of Cancellation or Expiration of Owner's Required Property Insurance. Within three (3) business days of the date the Owner becomes aware of an impending or actual cancellation or expiration of any property insurance required by the Contract Documents, the Owner shall provide notice to the Contractor of such impending or actual cancellation or expiration. Unless the lapse in coverage arises from an act or omission of the Contractor: (1) the Contractor, upon receipt of notice from the Owner, shall have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by either the Owner or the Contractor; (2) the Contract Time and Contract Sum shall be equitably adjusted; and (3) the Owner waives all rights against the Contractor, Subcontractors, and Sub-subcontractors to the extent any loss to the Owner would have been covered by the insurance had it not expired or been cancelled. If the Contractor purchases replacement coverage, the cost of the insurance shall be charged to the Owner by an appropriate Change Order. The furnishing of notice by the Owner shall not relieve the Owner of any contractual obligation to provide required insurance.

§ 11.3 Waivers of Subrogation

§ 11.3.1 The Owner and Contractor waive all rights against (1) each other and any of their subcontractors. sub-subcontractors, agents, and employees, each of the other; (2) the Architect and Architect's consultants; and (3) Separate Contractors, if any, and any of their subcontractors, sub-subcontractors, agents, and employees, for damages caused by fire, or other causes of loss, to the extent those losses are covered by property insurance required by the Agreement or other property insurance applicable to the Project, except such rights as they have to proceeds of such insurance. The Owner or Contractor, as appropriate, shall require similar written waivers in favor of the individuals and entities identified above from the Architect, Architect's consultants, Separate Contractors, subcontractors, and sub-subcontractors. The policies of insurance purchased and maintained by each person or entity agreeing to waive claims pursuant to this section 11.3.1 shall not prohibit this waiver of subrogation. This waiver of subrogation shall be effective as to a person or entity (1) even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, (2) even though that person or entity did not pay the insurance premium directly or indirectly, or (3) whether or not the person or entity had an insurable interest in the damaged property.

§ 11.3.2 If during the Project construction period the Owner insures properties, real or personal or both, at or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, to the extent permissible by such policies, the Owner waives all rights in accordance with the terms of Section 11.3.1 for damages caused by fire or other causes of loss covered by this separate property insurance.

Init

AIA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 11.4 Loss of Use, Business Interruption, and Delay in Completion Insurance

The Owner, at the Owner's option, may purchase and maintain insurance that will protect the Owner against loss of use of the Owner's property, or the inability to conduct normal operations, due to fire or other causes of loss. The Owner waives all rights of action against the Contractor and Architect for loss of use of the Owner's property, due to fire or other hazards however caused.

§11.5 Adjustment and Settlement of Insured Loss

§ 11.5.1 A loss insured under the property insurance required by the Agreement shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Section 11.5.2. The Owner shall pay the Architect and Contractor their just shares of insurance proceeds received by the Owner, and by appropriate agreements the Architect and Contractor shall make payments to their consultants and Subcontractors in similar manner.

§ 11.5.2 Prior to settlement of an insured loss, the Owner shall notify the Contractor of the terms of the proposed settlement as well as the proposed allocation of the insurance proceeds. The Contractor shall have 14 days from receipt of notice to object to the proposed settlement or allocation of the proceeds. If the Contractor does not object, the Owner shall settle the loss and the Contractor shall be bound by the settlement and allocation. Upon receipt, the Owner shall deposit the insurance proceeds in a separate account and make the appropriate distributions. Thereafter, if no other agreement is made or the Owner does not terminate the Contract for convenience, the Owner and Contractor shall exceute a Change Order for reconstruction of the damaged or destroyed Work in the amount allocated for that purpose. If the Contractor timely objects to either the terms of the proposed settlement or the allocation of the proceeds, the Owner may proceed to settle the insured loss, and any dispute between the Owner and Contractor arising out of the settlement or allocation of the proceeds shall be resolved pursuant to Article 15. Pending resolution of any dispute, the Owner may issue a Construction Change Directive for the reconstruction of the damaged or destroyed Work.

ARTICLE 12 UNCOVERING AND CORRECTION OF WORK

§ 12.1 Uncovering of Work

§ 12.1.1 If a portion of the Work is covered contrary to the Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Architect, be uncovered for the Architect's examination and be replaced at the Contractor's expense without change in the Contract Time.

§ 12.1.2 If a portion of the Work has been covered that the Architect has not specifically requested to examine prior to its being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, the Contractor shall be entitled to an equitable adjustment to the Contract Sum and Contract Time as may be appropriate. If such Work is not in accordance with the Contract Documents, the costs of uncovering the Work, and the cost of correction, shall be at the Contractor's expense.

§ 12.2 Correction of Work

§ 12.2.1 Before Substantial Completion

The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, discovered before Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

§ 12.2.2 After Substantial Completion

§ 12.2.2.1 In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of any applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of notice from the Owner to do so, unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. During the one-year period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty. If the Contractor fails to correct nonconforming Work within a reasonable time during

Init. 1

AIA Document A201™ - 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treatie Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

* that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.5.

§ 12.2.2 The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.

§ 12.2.3 The one-year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Section 12.2.

§ 12.2.3 The Contractor shall remove from the site portions of the Work that arc not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.

§ 12.2.4 The Contractor shall bear the cost of correcting destroyed or damaged construction of the Owner or Separate Contractors, whether completed or partially completed, caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.

§ 12.2.5 Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one-year period for correction of Work as described in Section 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

§ 12.3 Acceptance of Nonconforming Work

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

ARTICLE 13 MISCELLANEOUS PROVISIONS

§ 13.1 Governing Law

The Contract shall be governed by the law of the place where the Project is located, excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 15.4.

§ 13.2 Successors and Assigns

§ 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns, and legal representatives to covenants, agreements, and obligations contained in the Contract Documents. Except as provided in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

§ 13.2.2 The Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate the assignment.

§ 13.3 Rights and Remedies

§ 13.3.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights, and remedies otherwise imposed or available by law.

§ 13.3.2 No action or failure to act by the Owner, Architect, or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed upon in writing.

Init. 1

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA® Document is protected by U.S. Copyright Law and international Treaties. Unauthorized reproduction or distribution of this AlA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 13.4 Tests and Inspections

§ 13.4.1 Tests, inspections, and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules, and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections, and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections, and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made so that the Architect may be present for such procedures. The Owner shall bear costs of tests, inspections, or approvals that do not become requirements until after bids are received or negotiations concluded. The Owner shall directly arrange and pay for tests, inspections, or approvals where building codes or applicable laws or regulations so require.

§ 13.4.2 If the Architect, Owner, or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection, or approval not included under Section 13.4.1, the Architect will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection, or approval, by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Architect of when and where tests and inspections are to be made so that the Architect may be present for such procedures. Such costs, except as provided in Section 13.4.3, shall be at the Owner's expense.

§ 13.4.3 If procedures for testing, inspection, or approval under Sections 13.4.1 and 13.4.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary by such failure, including those of repeated procedures and compensation for the Architect's services and expenses, shall be at the Contractor's expense.

§ 13.4.4 Required certificates of testing, inspection, or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Architect.

§ 13.4.5 If the Architect is to observe tests, inspections, or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.

§ 13.4.6 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

§ 13.5 Interest

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at the rate the parties agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT

§ 14.1 Termination by the Contractor

§ 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 30 consecutive days through no act or fault of the Contractor, a Subcontractor, a Sub-subcontractor, their agents or employees, or any other persons or entities performing portions of the Work, for any of the following reasons:

- Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be .1 stopped;
- .2 An act of government, such as a declaration of national emergency, that requires all Work to be stopped;
- .3 Because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents; or
- .4 The Owner has failed to furnish to the Contractor reasonable evidence as required by Section 2.2.

§ 14.1.2 The Contractor may terminate the Contract if, through no act or fault of the Contractor, a Subcontractor, a Sub-subcontractor, their agents or employees, or any other persons or entities performing portions of the Work, repeated suspensions, delays, or interruptions of the entire Work by the Owner as described in Section 14.3, constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.

AIA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. **User Notes:**

(1848930370)

§ 14.1.3 If one of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work executed, as well as reasonable overhead and profit on Work not executed, and costs incurred by reason of such termination.

§ 14.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor, a Subcontractor, a Sub-subcontractor, or their agents or employees or any other persons or entities performing portions of the Work because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

§ 14.2 Termination by the Owner for Cause

§ 14.2.1 The Owner may terminate the Contract if the Contractor

- .1 repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
- .2 fails to make payment to Subcontractors or suppliers in accordance with the respective agreements between the Contractor and the Subcontractors or suppliers;
- .3 repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority; or
- .4 otherwise is guilty of substantial breach of a provision of the Contract Documents.

§ 14.2.2 When any of the reasons described in Section 14.2.1 exist, and upon certification by the Architect that sufficient cause exists to justify such action, the Owner may, without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' notice, terminate employment of the Contractor and may, subject to any prior rights of the surety:

- Exclude the Contractor from the site and take possession of all materials, equipment, tools, and .1 construction equipment and machinery thereon owned by the Contractor;
- .2 Accept assignment of subcontracts pursuant to Section 5.4; and
- .3 Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work.

§ 14.2.3 When the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.

§ 14.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case may be, shall be certified by the Initial Decision Maker, upon application, and this obligation for payment shall survive termination of the Contract.

§ 14.3 Suspension by the Owner for Convenience

§ 14.3.1 The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work, in whole or in part for such period of time as the Owner may determine.

§ 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay, or interruption under Section 14.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent

- that performance is, was, or would have been, so suspended, delayed, or interrupted, by another cause .1 for which the Contractor is responsible; or
- .2 that an equitable adjustment is made or denied under another provision of the Contract.

§ 14.4 Termination by the Owner for Convenience

§ 14.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.

§ 14.4.2 Upon receipt of notice from the Owner of such termination for the Owner's convenience, the Contractor shall .1 cease operations as directed by the Owner in the notice;

Init. 1

AIA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

- .2 take actions necessary, or that the Owner may direct, for the protection and preservation of the Work;
- .3 except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.

§ 14.4.3 In case of such termination for the Owner's convenience, the Owner shall pay the Contractor for Work properly executed; costs incurred by reason of the termination, including costs attributable to termination of Subcontracts; and the termination fee, if any, set forth in the Agreement.

ARTICLE 15 CLAIMS AND DISPUTES

§ 15.1 Claims

§ 15.1.1 Definition

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, a change in the Contract Time, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim. This Section 15.1.1 does not require the Owner to file a Claim in order to impose liquidated damages in accordance with the Contract Documents.

§ 15.1.2 Time Limits on Claims

The Owner and Contractor shall commence all Claims and causes of action against the other and arising out of or related to the Contract, whether in contract, tort, breach of warranty or otherwise, in accordance with the requirements of the binding dispute resolution method selected in the Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Contractor waive all Claims and causes of action not commenced in accordance with this Section 15.1.2.

§ 15.1.3 Notice of Claims

§ 15.1.3.1 Claims by either the Owner or Contractor, where the condition giving rise to the Claim is first discovered prior to expiration of the period for correction of the Work set forth in Section 12.2.2, shall be initiated by notice to the other party and to the Initial Decision Maker with a copy sent to the Architect, if the Architect is not serving as the Initial Decision Maker, Claims by either party under this Section 15.1.3.1 shall be initiated within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.

§ 15.1.3.2 Claims by either the Owner or Contractor, where the condition giving rise to the Claim is first discovered after expiration of the period for correction of the Work set forth in Section 12.2.2, shall be initiated by notice to the other party. In such event, no decision by the Initial Decision Maker is required.

§ 15.1.4 Continuing Contract Performance

§ 15.1.4.1 Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 14, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents.

§ 15.1.4.2 The Contract Sum and Contract Time shall be adjusted in accordance with the Initial Decision Maker's decision, subject to the right of either party to proceed in accordance with this Article 15. The Architect will issue Certificates for Payment in accordance with the decision of the Initial Decision Maker.

§ 15.1.5 Claims for Additional Cost

If the Contractor wishes to make a Claim for an increase in the Contract Sum, notice as provided in Section 15.1.3 shall be given before proceeding to execute the portion of the Work that is the subject of the Claim. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

§ 15.1.6 Claims for Additional Time

§ 15.1.6.1 If the Contractor wishes to make a Claim for an increase in the Contract Time, notice as provided in Section 15.1.3 shall be given. The Contractor's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.

Init.

(1848930370)

AIA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

§ 15.1.6.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated, and had an adverse effect on the scheduled construction.

§ 15.1.7 Waiver of Claims for Consequential Damages

The Contractor and Owner waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, 1 business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- damages incurred by the Contractor for principal office expenses including the compensation of .2 personnel stationed there, for losses of financing, business and reputation, and for loss of profit, except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 14. Nothing contained in this Section 15.1.7 shall be deemed to preclude assessment of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

§ 15.2 Initial Decision

§ 15.2.1 Claims, excluding those where the condition giving rise to the Claim is first discovered after expiration of the period for correction of the Work set forth in Section 12.2.2 or arising under Sections 10.3, 10.4, and 11.5, shall be referred to the Initial Decision Maker for initial decision. The Architect will serve as the Initial Decision Maker, unless otherwise indicated in the Agreement. Except for those Claims excluded by this Section 15.2.1, an initial decision shall be required as a condition precedent to mediation of any Claim. If an initial decision has not been rendered within 30 days after the Claim has been referred to the Initial Decision Maker, the party asserting the Claim may demand mediation and binding dispute resolution without a decision having been rendered. Unless the Initial Decision Maker and all affected parties agree, the Initial Decision Maker will not decide disputes between the Contractor and persons or entities other than the Owner.

§ 15.2.2 The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions: (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, (4) suggest a compromise, or (5) advise the parties that the Initial Decision Maker is unable to resolve the Claim if the Initial Decision Maker lacks sufficient information to evaluate the merits of the Claim or if the Initial Decision Maker concludes that, in the Initial Decision Maker's sole discretion, it would be inappropriate for the Initial Decision Maker to resolve the Claim.

§ 15.2.3 In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Decision Maker in rendering a decision. The Initial Decision Maker may request the Owner to authorize retention of such persons at the Owner's expense.

§ 15.2.4 If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of the request, and shall either (1) provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished, or (3) advise the Initial Decision Maker that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Initial Decision Maker will either reject or approve the Claim in whole or in part.

§ 15.2.5 The Initial Decision Maker will render an initial decision approving or rejecting the Claim, or indicating that the Initial Decision Maker is unable to resolve the Claim. This initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) notify the parties and the Architect, if the Architect is not serving as the Initial Decision Maker, of any change in the Contract Sum or Contract Time or both. The initial decision shall be final and binding on the parties but subject to mediation and, if the parties fail to resolve their dispute through mediation, to binding dispute resolution.

§ 15.2.6 Either party may file for mediation of an initial decision at any time, subject to the terms of Section 15.2.6.1.

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AlA® Document is protected by U.S. Copyright Law and International Treaties. Init. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

§ 15.2.6.1 Either party may, within 30 days from the date of receipt of an initial decision, demand in writing that the other party file for mediation. If such a demand is made and the party receiving the demand fails to file for mediation within 30 days after receipt thereof, then both parties waive their rights to mediate or pursue binding dispute resolution proceedings with respect to the initial decision.

§ 15.2.7 In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.

§ 15.2.8 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

§ 15.3 Mediation

§ 15.3.1 Claims, disputes, or other matters in controversy arising out of or related to the Contract, except those waived as provided for in Sections 9.10.4, 9.10.5, and 15.1.7, shall be subject to mediation as a condition precedent to binding dispute resolution.

§ 15.3.2 The parties shall endeavor to resolve their Claims by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration is stayed pursuant to this Section 15.3.2, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.

§ 15.3.3 Either party may, within 30 days from the date that mediation has been concluded without resolution of the dispute or 60 days after mediation has been demanded without resolution of the dispute, demand in writing that the other party file for binding dispute resolution. If such a demand is made and the party receiving the demand fails to file for binding dispute resolution within 60 days after receipt thereof, then both parties waive their rights to binding dispute resolution proceedings with respect to the initial decision.

§ 15.3.4 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

§ 15.4 Arbitration

§ 15.4.1 If the parties have selected arbitration as the method for binding dispute resolution in the Agreement, any Claim subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of the Agreement. The Arbitration shall be conducted in the place where the Project is located, unless another location is mutually agreed upon. A demand for arbitration shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the arbitration. The party filing a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded.

§ 15.4.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.

§ 15.4.2 The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

Init.

AlA Document A201™ – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 00:42:00 ET on 05/30/2019 under Order No.9759279263 which expires on 03/22/2020, and is not for resale. User Notes:

(1848930370)

38

§ 15.4.3 The foregoing agreement to arbitrate and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement, shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

§ 15.4.4 Consolidation or Joinder

Init.

1

§ 15.4.4.1 Subject to the rules of the American Arbitration Association or other applicable arbitration rules, either party may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation, (2) the arbitrations to be consolidated substantially involve common questions of law or fact, and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).

§ 15.4.4.2 Subject to the rules of the American Arbitration Association or other applicable arbitration rules, either party may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

§ 15.4.4.3 The Owner and Contractor grant to any person or entity made a party to an arbitration conducted under this Section 15.4, whether by joinder or consolidation, the same rights of joinder and consolidation as those of the Owner and Contractor under this Agreement.

39

As Of: 07/09/2019

Time: 14:50:52 Date: 07/03/2019 Page:

Accts Pay # Received Date Due	Vendor	Amount	Memo
19715 07/09/2019 07/09/2019 6904	A R C Architects Inc		P#54 Professional Services May 2019
594 76 62 00 Buildings & Structures - Pa	001 000 576 General Fund	108,632.73	P#54 Prof. Services 05/2019
9746 07/09/2019 07/09/2019 3647	Agrishop, Inc	33.04	Synthetic 2 - Stroke Oil
576 80 31 02 Oper Supplies - Parks	001 000 576 General Fund	33.04	Synthetic 2 - Stroke Oil
9711 07/09/2019 07/09/2019 7066	Apex Engineering	5,132.67	P#58 Farallone Ave Sewer Prof. Engineering Services May 2019
594 35 63 03 Project Engineering Sewer	432 000 594 Sewer Improve	5,132.67	P#58 Farallone Ave Sewer Prof. Engineering Services 05/2019
9652 07/09/2019 07/09/2019 7201	Audio Innovations	800.00	Sound For Fircrest Fun Days
573 90 49 01 Community Events	001 000 573 General Fund	800.00	Sound For Fircrest Fun Days
9752 07/09/2019 07/09/2019 7201	Audio Innovations	500.00	Sound For Car Show
573 90 49 01 Community Events	001 000 573 General Fund	500.00	Sound For Car Show
	Total Audio Innovations	1,300.00	
9668 07/09/2019 07/09/2019 6542	Beal, Louisa	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
9666 07/09/2019 07/09/2019 5805	CenturyLink	60.60	Long Distance Access And Usage June 2019
518 10 42 00 Communication - Non Dep	001 000 518 General Fund	60.60	Long Distance 06/2019
9767 07/09/2019 07/09/2019 7379	Chough, Kwang S.	120.00	Korean Interpreting (2 Hrs)
512 50 41 03 Prof Srvs - Interpreter	001 000 512 General Fund	120.00	Korean Interpreting (2 Hrs) 9Z0099999
9748 07/09/2019 07/09/2019 7151	City of Seattle	1,900.00	Global Positioning Subscriber (June 2019 - June 2020)
531 50 48 00 Rep & Maint - Storm	415 000 531 Storm Drain	475.00	Global Positioning Subscriber (June 2019 - June
534 50 48 01 Rep & Maint - Water Main	425 000 534 Water Fund (de	475.00	2020) Global Positioning Subscriber (June 2019 - June 2020)
535 50 48 00 Rep & Maint - Sewer Main	430 000 535 Sewer Fund (de	475.00	Global Positioning Subscriber (June 2019 - June
542 30 48 01 Rep & Maint - Street Maint	101 000 542 City Street Fund	475.00	2020) Global Positioning Subscriber (June 2019 - June 2020)
9659 07/09/2019 07/09/2019 3555	Code Publishing Co	717 10	Municipal Code Undate June 2019

19659 07/09/2019 07/09/2019 3555

Code Publishing Co

717.10 Municipal Code Update June 2019

City Of Fircrest MCAG #: 0583

City Of Fircrest MCAG #: 0583

As Of: 07/09/2019

Time: 14:50:52 Date: 07/03/2019 Page:

Accts Pay # Received Date Due	Vendor	Amount	Memo
511 60 49 03 Codification Costs	001 000 511 General Fund	717.10	Municipal Code Update 06/2019
19770 07/09/2019 07/09/2019 8542	Consolidated Supply Co.	281.40	SAA # 1719, 1720, 1721 & 1722 14'' Pipe Wrenches
534 80 35 00 Small Tools & Equip - Wat 535 80 35 00 Small Tools & Equip-Swr (425 000 534 Water Fund (de 430 000 535 Sewer Fund (de		SAA # 1719, 1720, 1721 & 1722 14" Pipe Wrenches SAA # 1719, 1720, 1721 & 1722 14" Pipe Wrenches
19662 07/09/2019 07/09/2019 3573	Copiers Northwest Inc	32.71	Copier Usage 06/04/19 - 06/14/19 Police
521 22 45 00 Oper Rentals - Copier - Pol	001 000 521 General Fund	32.71	Copier Usage 06/04/19 - 06/14/19 Police
19686 07/09/2019 07/09/2019 3573	Copiers Northwest Inc	446.64	Copier Usage May 2019 - City Hall, Court, Parks Rec, Public Works
 512 50 45 00 Oper Rentals - Copier - Coi 518 10 45 00 Oper Rentals - Copier - No 531 50 45 00 Oper Rentals - Copier - Sto 534 10 45 02 Oper Rentals - Copier - Wa 535 10 45 00 Oper Rentals - Copier - Str 542 30 45 00 Oper Rentals - Copier - Str 571 10 45 01 Oper Rentals - Copier - Rec 576 80 45 00 Oper Rentals - Copier - Par 	001 000 512 General Fund 001 000 518 General Fund 415 000 531 Storm Drain 425 000 534 Water Fund (de 430 000 535 Sewer Fund (de 101 000 542 City Street Fund 001 000 571 General Fund 001 000 576 General Fund	228.00 19.79 19.79 19.79 19.79 99.68	Ct 05/2019 CH 05/2019 PW 05/2019 PW 05/2019 PW 05/2019 PW 05/2019 Rec 05/2019 Parks 05/2019
	Total Copiers Northwest Inc	479.35	
19675 07/09/2019 07/09/2019 7802	Core & Main LP	120.94	Parts For Sewer Maintenance
535 50 31 01 Oper Supplies - Sewer Mai	430 000 535 Sewer Fund (de	120.94	Parts For Sewer Maintenance
19717 07/09/2019 07/09/2019 7802	Core & Main LP	417.00	Supplies For Alameda Water Services
534 50 31 01 Oper Supplies - Water Main	425 000 534 Water Fund (de	417.00	Supplies For Alameda Water Services
	Total Core & Main LP	537.94	
19755 07/09/2019 07/09/2019 2451	Cregeur, Todd	200.00	Refund Pool Rental Fee 6/08/19 - Pool Not Ready
362 50 00 04 Pool Rental Revenue	001 000 360 General Fund	-200.00	Refund Pool Rental Fee 6/08/19 - Pool Not Ready
19679 07/09/2019 07/09/2019 3588	Daily Journal Of Commerce	592.20	P#57 Alameda Grind & Overlay Request For Sealed Bids
595 10 63 06 Project Eng - Alameda Reg	101 000 594 City Street Fund	592.20	P#57 Alameda Grind & Overlay Request For Sealed Bids
19714 07/09/2019 07/09/2019 3589	Databar Inc	2,320.06	June 2019 Statement Production, 3 Inserts And Postage

City Of Fircrest

MCAG #: 0583

Time: 14:50:52 Date: 07/03/2019

Page: 3

As Of: 07/09/2019

Accts			
Pay # Received Date Due	Vendor	Amount	Memo
518 10 49 01 Town Topics/Citizen Comr	001 000 518 General Fund	66.09	06/2019 Town Topics Insert
531 50 42 01 Postage - Storm	415 000 531 Storm Drain		06/2019 Postage
531 50 49 06 Mailing Service - Storm	415 000 531 Storm Drain		06/2019 Statement Production
534 10 42 01 Postage - Water	425 000 534 Water Fund (de		06/2019 Postage
534 10 49 04 Printing & Binding - Water	425 000 534 Water Fund (de		06/2019 CCR & WUE Insert
534 10 49 06 Mailing Service - Water	425 000 534 Water Fund (de		06/2019 Statement Production
535 10 42 02 Postage - Sewer	430 000 535 Sewer Fund (de		06/2019 Postage
535 10 49 05 Mailing Service - Sewer	430 000 535 Sewer Fund (de		06/2019 Statement Production
542 80 31 04 Beautification-Supplies	101 000 542 City Street Fund		06/2019 Community Sponsorship Insert
19729 07/09/2019 07/09/2019 3589	Databar Inc	373.94	Separate Mailing Town Topics June 2019
518 10 49 01 Town Topics/Citizen Comr	001 000 518 General Fund	373.94	Separate Mailing TT 06/2019
	Total Databar Inc	2,694.00	
19676 07/09/2019 07/09/2019 8271	Davis, Lindsay M	30.08	Mileage And Meal Reimbursement - MRSC Seminar
514 23 43 00 Travel - Finance	001 000 514 General Fund	30.08	Mileage And Meal Reimbursement - MRSC Semina - L. Davis
19677 07/09/2019 07/09/2019 8271	Davis, Lindsay M	49.50	Gym Fees Reimbursement (Jan - Jun)
514 23 20 00 Personnel Benefits-Finance	001 000 514 General Fund	49.50	Gym Fees Reimbursement (Jan - Jun) - L. Davis
	Total Davis, Lindsay M	79.58	
19703 07/09/2019 07/09/2019 8004	Davis, Stephanie	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
19756 07/09/2019 07/09/2019 6995	Endicott, Cynthia J.	294.00	Silver Sneakers And Yoga May 2019
571 20 49 06 Instructor Fees	001 000 571 General Fund	294.00	Silver Sneakers & Yoga 05/2019
19759 07/09/2019 07/09/2019 4858	Ewing Irrigation Products Inc	88.08	Chalk For Fields
576 80 31 02 Oper Supplies - Parks	001 000 576 General Fund	88.08	Chalk For Fields
19761 07/09/2019 07/09/2019 8610	F & L Building Maintenance, LLC	2,080.00	June 2019 Janitorial Services (Partial Month)
518 30 41 01 Contract Maintenance	001 000 518 General Fund	2,080.00	June 2019 Janitorial Services (Partial Month)
19725 07/09/2019 07/09/2019 8614	Fair, Darwin James	369.94	Reimbursement For Tow In Error
521 22 49 00 Miscellaneous - Police	001 000 521 General Fund	369.94	Reimbursement For Tow In Error

As Of: 07/09/2019

City Of Fircrest

MCAG #: 0583

Time: 14:50:52 Date: 07/03/2019 Page:

Accts			
Pay # Received Date Due	Vendor	Amount	Memo
9651 07/09/2019 07/09/2019 3638	Fircrest Golf Club	1,287.78	Golf Tank Land Rental July 2019
534 10 45 01 Land Rental/Water Tank	425 000 534 Water Fund (de	1,287.78	Golf Tank Land Rental 07/2019
19663 07/09/2019 07/09/2019 6774	Greenleaf Landscaping 1 Inc	4,252.06	Monthly Landscape Service June 2019
518 30 41 01 Contract Maintenance 542 80 49 03 Beautification Services (con	001 000 518 General Fund 101 000 542 City Street Fund		Monthly Landscape Service 06/2019 Monthly Landscape Service 06/2019
19678 07/09/2019 07/09/2019 3690	Holroyd Co Inc	105.00	Dump Fees (Sidewalk Concrete)
542 30 47 01 Dumping Fees - Street	101 000 542 City Street Fund	105.00	Dump Fees (Sidewalk Concrete)
19764 07/09/2019 07/09/2019 3690	Holroyd Co Inc	105.00	Dump Fees (Sidewalk Concrete)
542 30 47 01 Dumping Fees - Street	101 000 542 City Street Fund	105.00	Dump Fees (Sidewalk Concrete)
	Total Holroyd Co Inc	210.00	
19724 07/09/2019 07/09/2019 3692	Home Depot Credit Services	118.93	Plant Food, Coolant, Clock And Toilet Auger
518 30 31 00 Oper Sup/Facilities 518 30 31 03 Oper Sup/PWF 542 80 31 01 Operating Supplies - St Bea 548 65 48 12 O & M - Street 548 65 48 13 O & M - Storm	001 000 518 General Fund 001 000 518 General Fund 101 000 542 City Street Fund 501 000 548 Equipment Ren 501 000 548 Equipment Ren	22.03 26.43 1.71	Toilet Auger Clock - PW Plant Food Coolant For Vactor Coolant For Vactor
548 65 48 14 O & M - Wtr/Swr	501 000 548 Equipment Ren	10.31	Coolant For Vactor
19658 07/09/2019 07/09/2019 6089	Kitsap County Sheriff's Office	19.36	EVOC Without PIT - J. Roberts
521 22 49 02 Reg & Tuition - Police	001 000 521 General Fund	19.36	EVOC Without PIT - J. Roberts
9754 07/09/2019 07/09/2019 8415	Lee, Sierra	144.00	Mens Basketball Scorekeeper (12 Hrs)
571 20 49 08 Adult Basketball	001 000 571 General Fund	144.00	Mens Basketball Scorekeeper (12 Hrs)
19672 07/09/2019 07/09/2019 8600	Legacy Tapping, Inc	2,961.26	12 X 4 Hot Tap For Whittier Park Sprinkler System
594 34 63 01 Other Improvements Water	426 000 594 Water Improven	2,961.26	12 X 4 Hot Tap For Whittier Park Sprinkler System
9727 07/09/2019 07/09/2019 8404	Linda Kaye Briggs	4,063.10	P#55 Capital Campaign June 2019 (21.167 Hrs Plus Writing / Editing Brochure)
594 76 62 00 Buildings & Structures - Pa	001 000 576 General Fund	4,063.10	P#55 Capital Campaign June 2019 (21.167 Hrs Plu Writing / Editing Brochure)
19706 07/09/2019 07/09/2019 2227	Linggi Jr, Albert J.	33.50	Library Reimbursement 1/2 Year

City Of Fircrest MCAG #: 0583

As Of: 07/09/2019

Time: 14:50:52 Date: 07/03/2019 Page:

Accts Pay # Received Date Due	Vendor	Amount	Memo
572 21 49 00 Library Services	001 000 572 General Fund	33.50	Library Reimbursement 1/2 Year
19709 07/09/2019 07/09/2019 8613	Long Painting	113,829.03	P#56 High Tank Painting Progress Billing #1 Through 6/30/19
594 34 63 01 Other Improvements Water	426 000 594 Water Improve	113,829.03	P#56 High Tank Painting Progress Through 6/30/19
19655 07/09/2019 07/09/2019 2796	Lovelace, Jim	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
19758 07/09/2019 07/09/2019 3791	Lowe's Company-#338954	22.46	Supplies For Wading Pool
576 20 31 03 Oper Supplies - Pool	001 000 576 General Fund	22.46	Supplies For Wading Pool
19762 07/09/2019 07/09/2019 6424	McDaniel, Matthew	1,205.36	Refund For Withdrawn Application
386 00 00 00 Refundable Deposits 586 00 00 00 Deposit Refunds	001 000 380 General Fund 001 000 580 General Fund		Refund For Withdrawn Application Refund For Withdrawn Application
19757 07/09/2019 07/09/2019 3841	Metropolitan Ballet of Tacoma	567.00	Ballet Instruction April - May 2019
571 20 49 06 Instructor Fees	001 000 571 General Fund	567.00	Ballet Instruction 04/2019 - 05/2019
19670 07/09/2019 07/09/2019 3882	Nellist, Angela	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
19736 07/09/2019 07/09/2019 3923	Orca Pacific Inc	482.19	Chlorine For Wells (135 Gallons)
534 80 31 03 Oper Supplies - Chlorine	425 000 534 Water Fund (de	482.19	Chlorine For Wells (135 Gallons)
19737 07/09/2019 07/09/2019 3923	Orca Pacific Inc	1,392.98	Chlorine For Pool (390 Gallons)
576 20 31 01 Pool Supplies-Chemicals	001 000 576 General Fund	1,392.98	Chlorine For Pool (390 Gallons)
19738 07/09/2019 07/09/2019 3923	Orca Pacific Inc	571.48	Chlorine For Pool (160 Gallons)
576 20 31 01 Pool Supplies-Chemicals	001 000 576 General Fund	571.48	Chlorine For Pool (160 Gallons)
19739 07/09/2019 07/09/2019 3923	Orca Pacific Inc	553.62	Chlorine For Pool (155 Gallons)
576 20 31 01 Pool Supplies-Chemicals	001 000 576 General Fund	553.62	Chlorine For Pool (155 Gallons)
19740 07/09/2019 07/09/2019 3923	Orca Pacific Inc	193.97	Service Call For Wading Pool Heater
576 20 48 00 Rep & Maint - Pool	001 000 576 General Fund	193.97	Service Call For Wading Pool Heater
19741 07/09/2019 07/09/2019 3923	Orca Pacific Inc	95.34	Supplies For Pool
576 20 31 03 Oper Supplies - Pool	001 000 576 General Fund	95.34	Supplies For Pool

As Of: 07/09/2019

Time: 14:50:52 Date: 07/03/2019 Page: 6

Accts Pay # Received Date Due	Vendor	Amount	Memo
19742 07/09/2019 07/09/2019 3923	Orca Pacific Inc		Chlorine And Hydrochloric Acid For Pool
576 20 31 01 Pool Supplies-Chemicals	001 000 576 General Fund	381.28	Chlorine And Hydrochloric Acid For Pool
19743 07/09/2019 07/09/2019 3923	Orca Pacific Inc	1,480.54	Chlorine (300 Gallons), Hydrochloric Acid, And Testing Supplies / Equipment For Pool
576 20 31 01 Pool Supplies-Chemicals	001 000 576 General Fund	1,480.54	Chlorine (300 Gallons), Hydrochloric Acid, And Testing Supplies / Equipment For Pool
19744 07/09/2019 07/09/2019 3923	Orca Pacific Inc	514.69	Chlorine And Testing Supplies / Equipment For Pool
576 20 31 01 Pool Supplies-Chemicals	001 000 576 General Fund	514.69	Chlorine And Testing Supplies / Equipment For Poo
19745 07/09/2019 07/09/2019 3923	Orca Pacific Inc	440.11	Chlorine And Hydrochloric Acid For Pool
576 20 31 01 Pool Supplies-Chemicals	001 000 576 General Fund	440.11	Chlorine And Hydrochloric Acid For Pool
	Total Orca Pacific Inc	6,106.20	
19766 07/09/2019 07/09/2019 3957	PC Budget & Finance	7,262.00	April 2019 Special Election Costs
514 40 41 01 Special Elections & Voter I	001 000 511 General Fund	7,262.00	April 2019 Special Election Costs
19669 07/09/2019 07/09/2019 1429	Pasic, Brian	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
19751 07/09/2019 07/09/2019 8615	Pegasus Puppet Theatre	300.00	Fun Days Entertainment
573 90 49 01 Community Events	001 000 573 General Fund	300.00	Fun Days Entertainment
19723 07/09/2019 07/09/2019 3955	Petrocard Systems Inc	246.65	Gas / Fuel June 2019
548 65 31 12 Street Gas 548 65 31 13 Storm Gas	501 000 548 Equipment Ren 501 000 548 Equipment Ren		Street 07/2019 Storm 07/2019
19769 07/09/2019 07/09/2019 3956	Petty Cash-Corcoran	87.75	Petty Cash 7/9/19
 517 90 31 01 Health Program - Supplies 518 10 42 01 Postage - Non-Dept 518 30 31 00 Oper Sup/Facilities 534 80 31 02 Oper Supplies - Water 542 80 31 01 Operating Supplies - St Bea 571 10 31 02 Senior Supplies 576 80 31 02 Oper Supplies - Parks 	001 000 517 General Fund 001 000 518 General Fund 001 000 518 General Fund 425 000 534 Water Fund (de 101 000 542 City Street Fund 001 000 571 General Fund 001 000 576 General Fund	13.95 14.82 2.99 7.48 21.00	Wellness Supplies Postage Facilities Supplies Operating Supplies Beautification Supplies Senior Supplies Parks Supplies
10726 07/00/2010 07/00/2010 4929	Protoct Vouth Sports		Deskenound Checks 2010 Seesenal Uines

19726 07/09/2019 07/09/2019 4828

City Of Fircrest

MCAG #: 0583

Protect Youth Sports

288.55 Background Checks - 2019 Seasonal Hires

As Of: 07/09/2019

Time: 14:50:52 Date: 07/03/2019 7 Page:

Pay # Received Date Due Vendor 518 11 41 00 Prof Svcs - Personnel 001 000 518 General Fund 19733 07/09/2019 07/09/2019 3986 Puget Sound Energy, BOT-01H 531 50 47 02 Public Utility Services/Bldi 415 000 531 Storm Drain 534 10 47 00 Utility Services/Building - 425 000 534 Water Fund (de 535 10 47 00 Utility Services/Building - 430 000 535 Sewer Fund (de 542 30 47 02 Electricity & Gas/Bldg - St 101 000 542 City Street Fund	41.65 10.41 10.41 10.41 10.42	Background Checks - 2019 Seasonal Hires Natural Gas - Public Works June 2019 Gas - PW 06/2019 Gas - PW 06/2019 Gas - PW 06/2019
19733 07/09/2019 07/09/2019 3986 Puget Sound Energy, BOT-01H 531 50 47 02 Public Utility Services/Blds 415 000 531 Storm Drain 534 10 47 00 Utility Services/Building - 425 000 534 Water Fund (de 535 10 47 00 Utility Services/Building - 430 000 535 Sewer Fund (de	41.65 10.41 10.41 10.41 10.42	Natural Gas - Public Works June 2019 Gas - PW 06/2019 Gas - PW 06/2019 Gas - PW 06/2019
531 50 47 02 Public Utility Services/Bldg 415 000 531 Storm Drain 534 10 47 00 Utility Services/Building - 3 425 000 534 Water Fund (de 535 10 47 00 Utility Services/Building - 3 430 000 535 Sewer Fund (de	10.41 10.41 10.41 10.42	Gas - PW 06/2019 Gas - PW 06/2019 Gas - PW 06/2019
534 10 47 00 Utility Services/Building - 425 000 534 Water Fund (de 535 10 47 00 Utility Services/Building - 430 000 535 Sewer Fund (de	10.41 10.41 10.42	Gas - PW 06/2019 Gas - PW 06/2019
535 10 47 00 Utility Services/Building - 430 000 535 Sewer Fund (de	10.41 10.42	Gas - PW 06/2019
	10.42	
542 50 47 02 Electricity & Gas/Bidg - St. 101 000 542 City Street Fund		$C \rightarrow DW 0 C 2010$
		Gas - PW 06/2019
19734 07/09/2019 07/09/2019 3986 Puget Sound Energy, BOT-01H	1,990.33	Natural Gas - Rec And Pool June 2019
576 20 47 00 Public Utility Services - Po 001 000 576 General Fund		Gas - Pool 06/2019
576 80 47 00 Public Utility Services - Pa001 000 576 General Fund	678.58	Gas - REC 06/2019
19735 07/09/2019 07/09/2019 3986 Puget Sound Energy, BOT-01H	39.27	Natural Gas - City Hall June 2019
518 30 47 00 Public Utility Services - Cit001 000 518 General Fund	39.27	Gas - CH 06/2019
Total Puget Sound Energy, BOT-01H	2,071.25	
19765 07/09/2019 07/09/2019 4683 Puyallup, City of	100.05	Jail Services May 2019
523 60 40 01 Jail 001 000 523 General Fund	100.05	Jail Services 05/2019
19708 07/09/2019 07/09/2019 5710 Rainier Connect, Mashell Telecom	106.95	Internet Access Fee July 2019
518 81 42 00 Communication - I/S 001 000 518 General Fund	106.95	Internet 07/2019
19705 07/09/2019 07/09/2019 6831 Reis, Doug	33.50	Library Reimbursement 1/2 Year
572 21 49 00 Library Services 001 000 572 General Fund	33.50	Library Reimbursement 1/2 Year
19774 07/09/2019 07/09/2019 4004 Reserve Account	2,000.00	#15690704 Meter Refill 06/24/19
518 10 42 01 Postage - Non-Dept 001 000 518 General Fund	2,000.00	#15690704 Meter Refill 06/24/19
19684 07/09/2019 07/09/2019 4026 S & B Inc	1,080.87	Drake Street And Princeton L/S System Repairs
535 50 48 00 Rep & Maint - Sewer Main 430 000 535 Sewer Fund (de	1,080.87	Drake Street And Princeton L/S System Repairs
19661 07/09/2019 07/09/2019 4035 Sarco Supply	83.66	Janitorial Supplies - Pool
576 20 31 02 Janitorial Supplies - Pool001 000 576 General Fund	83.66	Janitorial Supplies - Pool
19671 07/09/2019 07/09/2019 4035 Sarco Supply	49.67	Janitorial Supplies - Rec
518 30 31 01 Oper Sup/Rec Bldg 001 000 518 General Fund	49.67	Janitorial Supplies - Rec
19710 07/09/2019 07/09/2019 4035 Sarco Supply	19.76	Disposable Gloves

City Of Fircrest MCAG #: 0583

As Of: 07/09/2019

Time: 14:50:52 Date: 07/03/2019 Page: 8

Accts			
Pay # Received Date Due	Vendor	Amount	
518 30 31 04 Oper Sup/CH	001 000 518 General Fund	19.76	Disposable Gloves
19768 07/09/2019 07/09/2019 4035	Sarco Supply	76.82	Janitorial Supplies - Public Works
518 30 31 03 Oper Sup/PWF	001 000 518 General Fund	76.82	Janitorial Supplies - PW
	Total Sarco Supply	229.91	
19656 07/09/2019 07/09/2019 5728	Shelton*, Paul J	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
19653 07/09/2019 07/09/2019 8598	Sherman, Nancy	33.50	Library Reimbursement 1/2 Year
572 21 49 00 Library Services	001 000 572 General Fund	33.50	Library Reimbursement 1/2 Year
19660 07/09/2019 07/09/2019 4056	Sherwin-Williams Company	56.54	Paint For Tot Lot Doors
518 30 31 05 Oper Sup/Park Struct	001 000 518 General Fund	56.54	Paint For Tot Lot Doors
19704 07/09/2019 07/09/2019 3171	Shigeno, June	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
19750 07/09/2019 07/09/2019 4060	Sir Speedy	134.08	Fircrest Fun Days Posters
573 90 49 01 Community Events	001 000 573 General Fund	134.08	Fun Days Posters
19753 07/09/2019 07/09/2019 4060	Sir Speedy	120.89	Artwork For Car Show Banner And Shirts
573 90 49 01 Community Events	001 000 573 General Fund	120.89	Artwork For Car Show Banner And Shirts
	Total Sir Speedy	254.97	
19664 07/09/2019 07/09/2019 7308	SiteCrafting Inc	99.00	Monthly Hosting Cityoffircrest.net 6/15/19
518 81 41 02 Web Design & Maintenanc	001 000 518 General Fund	99.00	Cityoffircrest.net 6/15/19
19674 07/09/2019 07/09/2019 5602	Six Robblees' Inc	42.25	#54557D (2006) Dump Truck New Mirrors
548 65 48 12 O & M - Street	501 000 548 Equipment Ren	42.25	#54557D (2006) Dump Truck New Mirrors
19665 07/09/2019 07/09/2019 4065	Smith, Alling, P.S.	50.00	Judge Pro Tempore 06/20/19 (1 Hr)
512 50 41 02 Prof Srvs - Pro Temp Judge	001 000 512 General Fund	50.00	Judge Pro Tempore 06/20/19 (1 Hr)
19667 07/09/2019 07/09/2019 7728	Sorensen, Peter H	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year

City Of Fircrest MCAG #: 0583

As Of: 07/09/2019

City Of Fircrest

MCAG #: 0583

Time: 14:50:52 Date: 07/03/2019 Page: 9

ay # Received Date Due	Vendor	Amount	Memo
9749 07/09/2019 07/09/2019 4690	Sound Inspections LLC	1,540.60	Inspections, Mileage And Calls June 2019
524 20 41 01 Bldg Inspec/Plan Review	001 000 524 General Fund	1,540.60	Inspections, Mileage And Calls 06/2019
9680 07/09/2019 07/09/2019 4084	Staples Business Advantage	79.74	Supplies - Rec And Public Works
531 50 35 00 Small Tools & Equip - Stor	415 000 531 Storm Drain		Supplies - PW
534 10 35 00 Small Tools & Equip - Wat	425 000 534 Water Fund (de		Supplies - PW
535 10 35 00 Small Tools-Swr Admin	430 000 535 Sewer Fund (de		Supplies - PW
571 10 31 00 Office Supplies - Rec	001 000 571 General Fund	57.81	Supplies - Rec
9681 07/09/2019 07/09/2019 4084	Staples Business Advantage	57.69	Public Works Supplies
531 50 31 01 Office Supplies - Storm	415 000 531 Storm Drain		Supplies - PW
534 10 31 00 Office Supplies - Water	425 000 534 Water Fund (de		Supplies - PW
535 10 31 00 Office Supplies - Swr Adm	430 000 535 Sewer Fund (de	19.23	Supplies - PW
9682 07/09/2019 07/09/2019 4084	Staples Business Advantage	57.69	Public Works Supplies
531 50 31 01 Office Supplies - Storm	415 000 531 Storm Drain	19.23	Supplies - PW
534 10 31 00 Office Supplies - Water	425 000 534 Water Fund (de	19.23	Supplies - PW
535 10 31 00 Office Supplies - Swr Adm	430 000 535 Sewer Fund (de	19.23	Supplies - PW
9730 07/09/2019 07/09/2019 4084	Staples Business Advantage	239.74	Supplies - Central And Police
518 10 34 01 Central Office Supplies	001 000 518 General Fund	114.36	Central Supplies
521 22 31 00 Office & Oper Sup - Police	001 000 521 General Fund	125.38	Police Supplies
9731 07/09/2019 07/09/2019 4084	Staples Business Advantage	68.19	Central Supplies
518 10 34 01 Central Office Supplies	001 000 518 General Fund	68.19	Central Supplies
9732 07/09/2019 07/09/2019 4084	Staples Business Advantage	172.77	Court Supplies
512 50 31 00 Office & Oper Sup-Court	001 000 512 General Fund	172.77	Court Supplies
	Total Staples Business Advantage	675.82	
9654 07/09/2019 07/09/2019 6030	Storaasli, Justine	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
9747 07/09/2019 07/09/2019 4110	Superior Linen Service	157.72	Linen Service 6/06/19 And 6/20/19
576 80 49 00 Miscellaneous - Parks	001 000 576 General Fund	157.72	Linen Service 6/06/19 & 6/20/19
9763 07/09/2019 07/09/2019 4139	Tapco Visa Card	35.16	Network Adapter For C. Corcoran's Compute
518 81 35 00 Small Tools & Equip - I/S	001 000 518 General Fund	35.16	Network Adapter For C. Corcoran's Computer

As Of: 07/09/2019

City Of Fircrest

MCAG #: 0583

Time: 14:50:52 Date: 07/03/2019 Page:

10

Accts Pay # Received Date Due	Vendor	Amount	Memo
19673 07/09/2019 07/09/2019 5151	Tegco Fence Corporation		Entrance Gate Repair - Public Works
534 50 48 01 Rep & Maint - Water Main 535 50 48 00 Rep & Maint - Sewer Main 542 30 48 01 Rep & Maint - Street Maint	425 000 534 Water Fund (de 430 000 535 Sewer Fund (de 101 000 542 City Street Fund	195.11 195.11	Entrance Gate Repair - PW Entrance Gate Repair - PW Entrance Gate Repair - PW
19657 07/09/2019 07/09/2019 6278	Tucker, Ben	67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund	67.00	Library Reimbursement 1 Year
19771 07/09/2019 07/09/2019 8484	US Bank, Recreation Dept Account	19.00	Rec Charges Through 06/07/19
573 90 49 01 Community Events 576 20 31 00 Office Supplies - Pool	001 000 573 General Fund 001 000 576 General Fund		Poster Boards Laminated Pool Signs
19772 07/09/2019 07/09/2019 8484	US Bank, Recreation Dept Account	220.09	Rec Charges Through 6/24/19
571 10 31 00 Office Supplies - Rec 573 90 49 01 Community Events	001 000 571 General Fund 001 000 573 General Fund		SIM Card Strawberry Feed Supplies
19773 07/09/2019 07/09/2019 8484	US Bank, Recreation Dept Account	1,401.67	Rec Charges Through 6/20/19
517 90 31 01 Health Program - Supplies 573 90 49 01 Community Events	001 000 517 General Fund 001 000 573 General Fund		Wellness Lunch Supplies Strawberry Feed, Car Show And Fun Days Supplies
	Total US Bank, Recreation Dept Account	1,640.76	
19683 07/09/2019 07/09/2019 4177	University Place Radiator	82.49	#63582D Oil Change
548 65 48 12 O & M - Street	501 000 548 Equipment Ren	82.49	#63582D Oil Change
19718 07/09/2019 07/09/2019 4177	University Place Radiator	70.79	#63581D Oil Change
548 65 48 14 O & M - Wtr/Swr	501 000 548 Equipment Ren	70.79	#63581D Oil Change
	Total University Place Radiator	153.28	
19716 07/09/2019 07/09/2019 4178	University Place Refuse Inc	224.00	ROW Cleaning June 2019
542 30 47 01 Dumping Fees - Street	101 000 542 City Street Fund	224.00	ROW Cleaning 06/2019
19712 07/09/2019 07/09/2019 4179	Unum Life Insurance Company of America	46.80	Retired Benefits July 2019
521 22 20 02 LEOFF I Long Term Care]	001 000 521 General Fund	46.80	Police 0220603-011 07/2019
19728 07/09/2019 07/09/2019 4180	Utilities Underground	92.40	Locates June 2019
534 10 49 00 Miscellaneous - Water 535 10 49 00 Miscellaneous - Sewer	425 000 534 Water Fund (de 430 000 535 Sewer Fund (de		Locates 06/2019 Locates 06/2019

As Of: 07/09/2019

Time: 14:50:52 Date: 07/03/2019 Page:

11

ccts ay # Received Date Due	Vendor		Amount	Memo
9775 07/09/2019 07/09/2019 3645	WEX BANK, Wright Express FSC		2,742.07	Gas / Fuel June 2019
548 65 31 06 Facilities Gas 548 65 31 08 Police Gas 548 65 31 11 Parks/Rec Gas 548 65 31 11 Parks/Rec Gas 548 65 31 12 Street Gas 548 65 31 14 Wtr/Swr Gas	501 000 548 Equipment Re 501 000 548 Equipment Re	n n n	1,550.01 213.43 15.40 436.61	Facilities 06/2019 Police 06/2019 Parks 06/2019 Parks 06/2019 Street 06/2019 W / S 06/2019
9713 07/09/2019 07/09/2019 4229	Washington State Patrol		600.00	2nd Quarter 2019 ACCESS User Fee
521 22 41 03 WACIC/NCIC	001 000 521 General Fund		600.00	2nd Qtr 2019 ACCESS User Fee
9707 07/09/2019 07/09/2019 6666	Wilhelm, Robert		67.00	Library Reimbursement 1 Year
572 21 49 00 Library Services	001 000 572 General Fund		67.00	Library Reimbursement 1 Year
9760 07/09/2019 07/09/2019 5286	Winsupply		460.86	Irrigation Supplies For Fircrest Park
576 80 31 02 Oper Supplies - Parks	001 000 576 General Fund		460.86	Irrigation Supplies For Fircrest Park
9685 07/09/2019 07/09/2019 4273	Zumar Industries Inc		250.58	Vehicle Decals For #68579D Facilities Van And 2019 Freightliner Vactor Truck
594 48 64 06 Facilities - ERR Capital 594 48 64 12 Street - ERR Capital 594 48 64 13 Storm - ERR Capital 594 48 64 14 Water/Sewer - ERR Capita	501 000 548 Equipment Re 501 000 548 Equipment Re 501 000 548 Equipment Re 501 000 548 Equipment Re	n n	14.29 14.29	Vehicle Decals For #68579D Facilities Van Vehicle Decals For 2019 Freightliner Vactor Truck Vehicle Decals For 2019 Freightliner Vactor Truck Vehicle Decals For 2019 Freightliner Vactor Truck
E. J	Report 7	– Fotal:	284,698.79	
101 Ci 415 St 425 W 426 W 430 Se 432 Se	eneral Fund ty Street Fund orm Drain ater Fund (department) ater Improvement Fund wer Fund (department) wer Improvement Fund uipment Rental Fund	148,398.62 2,932.06 1,258.23 3,896.30 116,790.29 2,842.05 5,132.67 3,448.57		
This report has been reviewed by:		2,		

REMARKS:

City Of Fircrest

MCAG #: 0583

JUNE 17, 2019 FIRCREST CITY COUNCIL MEETING MINUTES – STUDY SESSION 1

CALL TO ORDER, PLEDGE OF ALLEGIANCE AND ROLL CALL

Mayor Hunter T. George called the special meeting to order at 6:00 P.M. and led the pledge of allegiance. Councilmembers David M. Viafore, Shannon Reynolds, Brett Wittner, Denny Waltier, Blake Surina, and Jamie Nixon were present.

AGENDA MODIFICATIONS

There were none.

2020 BUDGET EXPECTATIONS DISCUSSION

City Manager Pingel requested feedback from Council regarding 2020 budget expectations beyond maintaining a status quo budget, the pool and bathhouse construction, completing the community center design, parks projects associated with the bond, and software upgrades. Viafore commented on his support for the water meter upgrade and requested staff to research grant opportunities to fund the water meter upgrade. Viafore requested staff to solicit interest from part-time employees on benefits to determine the need. Viafore requested funds for emphasis patrol, particularly for traffic and truck enforcement. Reynolds commented on her support for a cyber security audit to identify gaps, and data architecture maps. Pingel indicated the City was in the process of obtaining a high-level cyber security scan and would report back on the results. Wittner commented on his support for the emphasis patrols and water meter upgrades, looking for funding for an additional police officer and pedestrian crossing safety improvements, and hiring a custodian instead of contracting those services. Waltier commented on his support for pedestrian crossing safety improvements, water meter upgrade phased-in plan, and emphasis patrol, and requested an electronic reader board at Alice Peers Park. Surina requested to include usage on the water utility electronic bill and to take efforts to encourage customers to convert from paper to electronic utility billing and use those postage savings to survey residents on pertinent topics. Surina requested staff to research a facility assessment for the police station to determine costs and options and funding set aside for the start-up of civic groups. Surina inquired on the funding status of the Parks and Recreation storage shed; Pingel indicated it was sufficiently budgeted in this year's budget and that staff was currently reviewing options to fulfill the need. Surina requested funds to evaluate the bicycle lanes and tie into the greater bicycle system. Nixon inquired on the water meter upgrade; Public Works Director Wakefield briefed the Council on the Department's recommendation for advanced metering infrastructure and a phased-in replacement approach. Nixon requested more information on emphasis patrol and determining the need for an additional officer, and commented on his support for the water meter upgrade, part-time benefits, cyber security. George commented on his support of the councilmembers' requests, and requested funding for the recreation registration software. George requested staff to consider submitting a legislative capital budget request for the water meter upgrade and pedestrian crossing safety improvements and a federal grant application for the additional police officer. There was a brief discussion on the Facilities Department and custodial services. Viafore inquired if the City would consider purchasing the vacant 9-acre lot on Mildred Street to be eligible to apply for Brownfield grants. Reynolds recommended adding an analyst to staff in the future to assist the City.

JUNE 17, 2019 FIRCREST CITY COUNCIL MEETING MINUTES – STUDY SESSION 2

ELECTRON WAY PARKING EVALUATION

Wakefield briefed the Council on the Electron Way parking evaluation and preferred channelization concept, which included reviewing opportunities for additional parking and traffic calming features along Electron Way adjacent to Fircrest Park. Wakefield indicated that the proposed concept would create approximately 17 additional parking stalls in front of the park and that the 45-degree angled parking with 9' wide parking stalls would allow for ease of access for drivers with expected high stall turnover rates. Wakefield stated that the proposed striping concept could be implemented without any major construction and be completed during 2019. Wakefield indicated this would be a \$30,000 project that could be completed via contract and that the maintenance would be done in-house. After a brief discussion on the parking concept, roundabouts, handicap parking, the need for additional parking, and maintenance costs, there was a general consensus to remove the two proposed parking stalls nearest Contra Costa Avenue, widen the parking stalls greater than the standard 9' x 18', and to initiate work now for completion later this year. Wakefield indicated he would return to Council at a later date with a modified concept and a plan to have the proposed parking completed in 2019.

ADJOURNMENT

Reynolds MOVED to adjourn the meeting at 7:22 P.M., seconded by Wittner. <u>The Motion</u> <u>Carried (7-0).</u>

Hunter T. George, Mayor

Jessica Nappi, City Clerk

CALL TO ORDER, PLEDGE OF ALLEGIANCE, AND ROLL CALL

Mayor Hunter T. George called the regular meeting to order at 7:01 P.M. and led the Pledge of Allegiance. Councilmembers Shannon Reynolds, Brett Wittner, Denny Waltier, Blake Surina, and Jamie Nixon were present. George indicated Councilmember David M. Viafore had notified him that he would be late and was excused.

PRESIDING OFFICER'S REPORT

A. Legislative Delegation Session Report

George presented Senator O'Ban and Representatives Kilduff and Leavitt briefed who the Council on the 2019 legislative session report, which included topics on mental health and substance abuse disorder; Western State Hospital; public safety; homelessness; expansion of property tax exemptions; special education funding; health care billing; public infrastructure; supporting local community and school projects; supporting veterans, service members, and families; car tab tax reduction; medical and consumer debt; and workforce development.

Viafore arrived at 7:19 P.M.

Council thanked the legislative delegation for their support on the capital budget, which included \$1 million for the Fircrest pool and community center project. There was a brief discussion on the homestead exemption, the location of the new mental health facilities and their impacts to the nearby communities and schools, restructuring the state's tax system, tax reduction, and health care affordability. Council thanked O'Ban, Kilduff, and Leavitt for their report.

At the request of Waltier, the meeting recessed at 8:01 P.M. for a five-minute recess. The meeting reconvened at 8:06 P.M.

B. Resolution of Appreciation for Karen Patjens

George briefed the Council on Planning Commissioner Patjens' twenty years service to the Planning Commission and read the resolution of appreciation into the record. Viafore **MOVED to approve Resolution No. 1592, recognizing Karen Patjens for twenty years of dedicated service to the City of Fircrest Planning Commission; seconded by Nixon.** George invited councilmember comment; Viafore, Surina, and George thanked her for her service and commitment. George invited public comment; Planning and Building Administrator Stahlnecker commented on her appreciation of Patjens's contribution to the Planning Commission. Patjens thanked the City for the opportunity to serve and commented on her appreciation of the acknowledgements. <u>The Motion Carried (7-0)</u>.

C. Community Center and Pool Project

City Manager Pingel reported on the community center and pool bids received on June 25, 2019, stating four bids were received and ranged from \$5.0 to \$5.6 million and that due diligence would be done before a staff recommendation would be brought before Council. Pingel indicated this item would be brought before Council for discussion at the July 9, 2019 regular meeting and for approval at the July 23, 2019 regular meeting. There was a brief discussion on process and next steps, documentation, and communicating to the public on the shortened swim season. Parks and Recreation Director Grover indicated August 18, 2019 has been communicated as the scheduled last date for the pool season. George reported that a grant

application was submitted with the Cheney Foundation and Pingel indicated he would include the Foundation's application process with the weekly update.

George read into the record a thank you note from Mary Lewis regarding the City's appearance and flower baskets, and commented on thank you note from Tacoma Public Utilities Regional Relations Manager, LaTasha Wortham. George reported that he and Pingel had recently met with Tacoma School District Executive Director of Planning & Construction and Transportation, Morris Aldridge, and Parametrix Project Manager, Jim Dugan, regarding the school district's bond package for February 2020. George indicated they were developing a list of projects and stated the replacement of Whittier Elementary School was on the proposed list. George indicated the proposed list would be presented to the school board in July 2019 and that the school district representatives would present to Council after the July 2019 school board meeting. Viafore requested staff to research what could be built on the existing site and ensure that the City's codes were reflective of the City's vision. Surina inquired if there were opportunities to partner with Tacoma on the field maintenance and improvements.

PUBLIC COMMENTS FOR ITEMS NOT ON THE AGENDA

George invited public comment; there were none.

COMMITTEE, COMMISSION, AND LIAISON REPORTS

A. Parks and Recreation

Waltier had no report to provide and commented on the recent Strawberry Feed event and swimming lessons. Grover reported that swimming lessons were nearly sold out and that there were 90 registrations for the swim team.

B. Pierce County Regional Council

Reynolds had no report to provide as Pierce County Regional Council meeting was canceled.

C. Public Safety, Courts

Surina reported on the safety glass installation, police presence at the community center, a July 10, 2019 workshop for the Police Department on carrying and administering Narcan, and police vehicle laptop replacements. Viafore commented on looking into whether the leftover safety glass could be used.

D. Street, Water, Sewer, and Storm Drain

Nixon reported that the bids for the Alameda Grind and Overlay project would be opened on June 28, 2019 and be brought before Council for an award on July 9, 2019. Nixon reported on the PSE work on Alameda Avenue and the status of the High Tank painting project, and stated that construction on the Farallone Avenue sewer connection project would begin late summer 2019. Nixon indicated that the water main projects on the 400 block of Summit Avenue and 600 block of Contra Costa Avenue would begin after July 4, 2019. Nixon indicated that the crack sealing, street painting and completion of LED conversion would be completed late summer. Nixon reported that the water use efficiency and consumer confidence report was included in the current recent utility bill. Nixon reported that as part of water conservation efforts, City services include landscape moisture readings for residents and that a water conservation booth would be included at the upcoming Fun Days event. There was a brief discussion on the LED conversion project and related cost savings, and Viafore requested staff to look into the road service at the 1200 block of Farallone Avenue.

E. Other Liaison Reports

Reynolds commented available grants through Pool Safely Grant Program, which provides local governments with assistance to help implement enforcement and education programs, with the goal of preventing drownings in pools.

CONSENT CALENDAR

George requested the City Clerk read the Consent Calendar as follows: approval of Voucher No. 213204 through Voucher Check No. 213261 in the amount of \$302,433.52; approval of Payroll Check No. 13663 through Payroll Check No. 13668 in the amount of \$115,979.28; approval of the May 28, 2019 regular meeting minutes as corrected; and approval of the June 11, 2019 Regular meeting minutes. George briefed the Council on the corrections made to the May 28, 2019 regular meeting minutes. **Viafore MOVED to approve the Consent Calendar as corrected and as read; seconded by Wittner**. <u>The Motion Carried As Corrected and As Read (7-0)</u>.</u>

PUBLIC HEARING

None scheduled.

UNFINISHED BUSINESS

There was none.

NEW BUSINESS

A. Resolution: Fun Days Fireworks Display Agreement

Grover briefed the Council on a proposed resolution that would authorize an agreement with Alpha Pyrotechnics to provide a firework display for the City's July 20, 2019 Fun Days event. **Reynolds MOVED to adopt Resolution No. 1593, authorizing the City Manager to enter into agreement with Alpha Pyrotechnics for a firework display on July 20, 2019; seconded by Wittner**. George invited councilmember comment; none were provided. George invited public comment; none were provided. <u>The Motion Carried (7-0)</u>.

B. Resolution: National Night Out Fireworks Display Agreement

Police Chief Cheesman briefed the Council on a proposed resolution that would authorize an agreement with Alpha Pyrotechnics to provide a firework display for the City's August 6, 2019 National Night Out event. Cheesman indicated the cost of the fireworks display was \$6,700 and would be paid for with donated funds. Reynolds MOVED to adopt Resolution No. 1594, authorizing the City Manager to execute an agreement with Alpha Pyrotechnics, for a fireworks display on August 6, 2019; seconded by Nixon. George invited councilmember comment; Viafore inquired on the contract payment terms. Staff and City Attorney Smith confirmed payment was customarily made after the event. George invited public comment; none were provided. <u>The Motion Carried (7-0)</u>.

C. Resolution: Pierce County Brine Agreement

Pingel briefed the Council on the proposed amendment to the memorandum of agreement between Pierce County and the City of Fircrest for the purchase of road supplies primarily brine used for deicing the roads, stating this amendment amends the agreement due to changes in Pierce County's compensation and billing procedures. Reynolds MOVED to adopt Resolution No. 1595, authorizing the City Manager to execute a second amendment to the Memorandum of Agreement CC-91901 between Pierce County and the City of Fircrest; seconded by Wittner. George invited councilmember comment; Nixon commented on a concern from the Public Works Director regarding potential supply shortages at the county level. George invited public comment; none were provided. <u>The Motion Carried (7-0)</u>.

CITY MANAGER COMMENTS

Pingel reported that the City was in process of transitioning to Office 365 and would send instructions to Council on the Office 365 Outlook web app for access. Pingel requested Council feedback regarding the Kokich Larry Anderson statues, stating Marilyn Kokich's preference was for the City to accept the baseball statue and place it in the park overlooking the baseball fields. After a brief discussion on the need to develop a policy on receiving donated art and decommissioning, there was a general consensus to accept the baseball statue. **Viafore MOVED to authorize the City Manager to receive a donated sculpture from the Kokich family; seconded by Reynolds**. George invited councilmember comment; Wittner inquired if the motion should specify the baseball statue, there was a possibility the baseball statue would not be available and so the Council would still prefer to accept a statue. Council requested staff to communicate on behalf of Council its appreciation of Kokich's generosity. George invited public comment; none were provided. <u>The Motion Carried (7-0)</u>.

Viafore inquired on the status of the IT security scan; Pingel indicated the report would be completed within the next few weeks.

DEPARTMENT HEAD COMMENTS

None were provided.

COUNCILMEMBER COMMENTS

- Viafore thanked employees and management team for their City beautification efforts.
- Reynolds reported that she would be on vacation September 1-6, 2019.
- Wittner commented on the Strawberry Feed event.
- Waltier commented on the community sponsorship donation form included in the recent Town Topics issue and on the Regence Boulevard island status.
- Surina welcomed Waltier to his neighborhood.
- Nixon thanked the audience for their attendance and indicated he would be absent from the July 9, 2019 regular Council meeting.
- George requested councilmembers to provide staff notice of any upcoming meeting absences and vacation schedules, and reported that he would be absent from the July 9, 2019 regular Council meeting. George thanked the audience for their attendance.

EXECUTIVE SESSION

At 9:05 P.M., George reported that Council would take a three-minute recess and convene into Executive Session, not to exceed the hour of 9:50 P.M., to discuss the performance of a public employee pursuant to RCW 42.30.110. George noted that Pingel would be joining the Executive Session and that no further action would take place other than adjournment.

The Council reconvened into regular session at 9:26 P.M.

ADJOURNMENT Wittner MOVED to adjourn the meeting at 9:26 P.M., seconded by Reynolds. <u>The Motion</u> <u>Carried (7-0)</u>.

Hunter T. George, Mayor

Jessica Nappi, City Clerk

CALL TO ORDER, PLEDGE OF ALLEGIANCE AND ROLL CALL

Mayor Hunter T. George called the special meeting to order at 4:00 P.M. and led the pledge of allegiance. Councilmembers David M. Viafore, Shannon Reynolds, Brett Wittner, Denny Waltier, and Jamie Nixon were present. Councilmember Blake Surina was absent.

AGENDA MODIFICATIONS

There were none.

FIRCREST COMMUNITY POOL AND BATHHOUSE PROJECT

City Manager Pingel reported on the Fircrest Community Pool and Bathhouse project, stating the project was advertised for bids on May 30, 2019, and bids were opened on June 25, 2019. Pingel indicated that after legal consultation, it was recommended that the City reject all bids for the project in order to clarify the bid instructions to bidders and then re-bid the project. Pingel stated RCW 35.23.352 provided the City the right to reject any and all bids for any or no reason. Pingel indicated that a 13-day bid advertisement timeline was required per RCW 35.23.352 and stated that should the Council approve the recommended resolution, the bid would be posted July 2, 2019 with a bid opening on July 17, 2019. Pingel indicated that there would be an opportunity to discuss the project and contract at the July 9, 2019 regular Council meeting and July 15, 2019 study session, and that a recommendation for project bid award and contract approval would be brought to Council for consideration for approval at the July 23, 2019 regular Council meeting. Nixon MOVED to approve Resolution No. 1596, to reject all bids for the Fircrest Community Pool and Bathhouse project; seconded by Reynolds. George invited councilmember comment; Waltier inquired if staff had received responses from bidders regarding the bid readvertisement. Pingel indicated he had not. Viafore inquired about the clarifications; Pingel indicated the bid document instructions would clarify what would be required from bidders regarding bid submittals. George commented on the timeline and indicated a special meeting could be called should it be needed. Waltier inquired if the bidders were aware that Council would be taking action to reject all bids; Pingel indicated the bidders had been notified after the July 1, 2019 special meeting had been called. Wittner inquired on the likelihood of not receiving bids; Pingel indicated there was an interest in the project and that should no bids be submitted, the City would need to extend the bid date. George invited public comment; none were provided. The Motion Carried (6-0 with Surina absent).

ADJOURNMENT

Reynolds MOVED to adjourn the meeting at 4:08 P.M., seconded by Nixon. <u>The Motion</u> <u>Carried (6-0 with Surina absent)</u>.

Hunter T. George, Mayor

Jessica Nappi, City Clerk

NEW BUSINESS:Alameda Grind and Overlay Project Bid Award and ContractITEM 10A.Approval

FROM: Jerry Wakefield, Public Works Director

RECOMMENDED MOTION: I move the City Council authorize the City Manager to award the contract for the Alameda Avenue Grind and Overlay project to Tony Lind Paving, LLC in the amount of \$138,793.64 to prepare and execute the required contract documents, and to give the "Notice to Proceed" to commence construction of the project.

PROPOSAL: It is proposed that the City of Fircrest award the contract for the Alameda Avenue Grind and Overlay project to Tony Lind Paving, LLC in the amount of \$138,793.64. By this action, the Council gives authorization to prepare and execute contract documents, and give the low bidder a "Notice to Proceed" with the work detailed in the plans and specifications.

FISCAL IMPACT: Three (3) bids were received, which ranged from \$138,793.64 to \$179,375.00. The engineer's estimate was \$153,490.00. The project is being funding by a grant from the Puget Sound Regional Council (PSRC) and administered by WSDOT. The amount authorized for the construction phase of the contract is \$153,490.00.

ADVANTAGE: This bid is within the budget and grant amounts for this project. The low bidder is responsive and comes recommended to do this project.

DISADVANTAGES: None identified.

ALTERNATIVES: Not to proceed with the project and return funding to PSRC.

HISTORY: This project received the PSRC grant based on the scope and estimated costs prepared in December 2016. Due to delays incurred by PSRC, this project was finally obligated for funding in 2019. The City received the obligation for funding the Preliminary Engineering in January 2019. KPG was selected as the consultant to prepare the design and construction management of the project in March 2019. The project was authorized for construction in June 2019. Three bids were received on June 28, 2019. All bids were reviewed and considered responsive.

KPG tabulated the bids and checked on the responsiveness and references of the bids. Attached is a copy of KPG letter of recommendation to award to Tony Lind Paving. KPG and WSDOT acknowledges that Tony Lind Paving is a qualified bidder and that this is a responsive bid for the Alameda Avenue Grind and Overlay project.

ATTACHMENTS: <u>Certified Bid Tabulation</u> <u>KPG Recommendation Letter</u>



City of Fircrest Alameda Ave Grind & Overlay Project STPUP-2991(003)

I reviewed the bid tabs	from balanced	bids
XD and	a	6-28-19

Daniel Clark, KPG

	100						Tony Lin	d Paving, LLC		Paving and uction, Inc.	Miles Res	ources, LLC
Bid Item	Section		Quantity	Unit	Unit Cost	Base Cost	Unit Cost	Base Cost	Unit Cost	Base Cost	Unit Cost	Base Cost
		SECTION 1 PREPARATIO	N	1900 °, 19								Charles State
1		Minor Changes	1	EST	\$ 4,000.00	\$ 4,000.00	\$4,000.00	\$4,000.00	\$4,000.00	\$4,000.00	\$4,000.00	\$4,000.0
2		SPCC Plan	1	LS	\$ 1,000.00	\$ 1,000.00	\$720.00	\$720.00	\$500.00	\$500.00	\$975.00	\$975.0
3		Mobilization	1	LS	\$ 16,000.00	\$ 16,000.00	\$18,396.04	\$18,396.04	\$35,285.00	\$35,285.00		
4	1-10	Project Temporary Traffic Control	1	LS	\$ 20,000.00	\$ 20,000.00	\$23,184.00	\$23,184.00	\$22,500.00	\$22,500.00	\$43,050.00	\$21,600.0
5	2-02	Adjust Water Valve to Finished Grade	4	EA	\$ 750.00	\$ 3,000.00	\$780.00	\$3,120.00	\$1,000.00	\$4,000.00	-	\$43,050.0
6	2-02	Adjust Sewer Manhole to Finished Grade	2	EA	\$ 1,500.00	\$ 3,000.00	\$1,000.00	\$2,000.00	\$1,000.00	\$2,000.00	\$700.00 \$1,100.00	\$2,800.0
7	2-02	Adjust Catch Basin to Finished Grade	1	EA	\$ 1,250.00		\$1,000.00	\$1,000.00	\$800.00	\$800.00		\$2,200.0
8	4-04	Crushed Surfacing Base Course	25	TON	\$ 50.00	\$ 1,250.00	\$6.00	\$150.00	\$50.00		\$1,100.00	\$1,100.0
9	5-04	Pavement Repair Excavation Incl. Haul	25	CY	\$ 110.00		\$24.99	\$624.75	\$200.00	\$1,250.00	\$50.00	\$1,250.0
10	5-04	Planing Bituminous Pavement	3380	SY	\$ 5.00	\$ 16,900,00	\$6.10	\$20,618.00	\$6.00	\$5,000.00	\$250.00	\$6,250.0
11	5-04	HMA CI. 1/2 In. PG 64-22	420	TON	\$ 140.00	\$ 58,800.00	\$104.73	\$43,986.60		\$20,280.00	\$5.60	\$18,928.0
12	5-04	HMA CI. 1/2 In. PG 64-22 for Pavement Repair	25	TON	\$ 140.00	\$ 3,500.00	\$104.73	\$2,618.25	\$140.00	\$58,800.00	\$109.50	\$45,990.0
13	7-05	Supply Sewer Manhole Ring and Cover	2	EA	\$ 2,000.00		\$442.20		\$300.00	\$7,500.00	\$160.00	\$4,000.0
14	8-01	Erosion/Water Pollution Control		LS	\$ 2,500.00	\$ 2,500.00	\$2,976.00	\$884.40	\$1,000.00	\$2,000.00	\$350.00	\$700.0
15	8-01	Inlet Protection	11	EA	\$ 125.00		\$50.00	\$2,976.00	\$1,500.00	\$1,500.00	\$3,000.00	\$3,000.00
16	8-20	Traffic Induction Loop	3	EA	\$ 2,500.00	,		\$550.00	\$100.00	\$1,100.00	\$85.00	\$935.00
17	8-21	Paint Line	1020	LF	\$ 2,000	,	\$1,800.00	\$5,400.00	\$1,650.00	\$4,950.00	\$1,800.00	\$5,400.0
18	8-22	Plastic Stop Line	15	LF	\$ 25.00		\$6.00	\$6,120.00	\$5.50	\$5,610.00	\$0.50	\$510.00
19		Plastic Traffic Letter	17	EA			\$42.00	\$630.00	\$40.00	\$600.00	\$10.00	\$150.00
10.02			1/1	EA	\$ 250.00	\$ 4,250.00	\$106.80	\$1,815.60	\$100.00	\$1,700.00	\$150.00	\$2,550.00
			ESTIMATED C	ONSTRU	CTION COST	A150 400 00						
			LOTIMATED		ngency (xxx)	\$153,490.00		\$138,793.64		\$179,375.00		\$165,388.00
		τοτ	L ESTIMATED C			\$0.00	r					
		1017	LOTIMATED	UNSTRUC		\$153,490.00						

Bid Tab Analysis June 28, 2019



July 01, 2019

Jerry Wakefield, PE Director of Public Works City of Fircrest 115 Ramsdell St Fircrest, WA 98466

RE: Alameda Ave Grind & Overlay Project WSDOT Fed Aid No.: STPUP-2991(003)

Dear Jerry,

On June 28, 2019, sealed bids were received and opened for the Alameda Avenue Grind & Overlay Project in City of Fircrest City Hall at 10:00 a.m. Three bids were submitted, with the results of the bids and Engineer's Estimate as follows:

Bidder's Name	Bid	
Engineer's Estimate	\$153,490.00	
Tony Lind Paving, LLC	\$138,793.64	
Puget Paving and Construction, Inc.	\$179,375.00	
Miles Resources, LLC	\$165,388.00	

Tony Lind Paving, LLC is the apparent low bidder.

Contractor Qualifications

- Verified Bidder Responsibility Checklist and project documents.
- Exhibit 1: Bid Tabulations indicate that Tony Lind Paving, LLC has prepared the proposal correctly, without errors and projects as the low bid.
- Exhibit 2: Verification of Tony Lind Paving, LLC contractor's license from the Department of Labor and Industries; verification of Business Registrations; verification of Insurance of Coverage, verification of State Excise Tax Registration, verification that Tony Lind Paving, LLC is not on the State Tax Payment delinquent list.
- Exhibit 3: Tony Lind Paving, LLC is not on the State L&I and Federal "Debarred Contractors" List; verification that Tony Lind Paving, LLC is not on the State L&I Contractor Strike List.



- Exhibit 4: Reference Checks
 - City of Kirkland Ralph Utley "They do a lot of our grind and overlays and we have always been very pleased with their work. Paperwork has never been an issue for them"
 - Boeing Chet Williams "We have worked with them on a lot of Boeing work and with municipalities and they do excellent work and we have never had a problem with them in any way."
 - City of Sumner Andrew Leach "They worked for us a couple years ago and I remember they were pretty good and we had no problems. We even had some adds on that project and they were good to work with. The project was not Federally Funded."
- Exhibit 5: Subcontractor verification

Based on the outcome of the bids and a check of the Contractor's qualifications, I recommend that the City of Fircrest award a construction contract to Tony Lind Paving, LLC in the amount of \$138,793.64.

Please call me with any questions, (253) 777-5905

Sincerely,

Dove af

Daniel Clark, KPG Project Manager

Attachments: Bidder Responsibility Checklist and Project Documents

Exhibits 1-5

FEDERAL AID NO. STPUL-2991 (003) – ALAMEDA AVENUE GRIND & OVERLAY

BIDDER'S CHECKLIST

Bidder must execute and return with submittal:

- 1. NON-COLLUSION DECLARATION (WSDOT FORM 272-036I)
- 2. J BID PROPOSAL
- 3. <u>LOCAL AGENCY PROPOSAL SIGNATURE PAGE (WSDOT FORM 272-036K)</u>
- 4. LOCAL AGENCY PROPOSAL BOND (WSDOT FORM 252-001A)
- 5. <u>LOCAL AGENCY CERTIFICATION FOR FEDERAL AID CONTRACTS</u>
- 6. <u>LOCAL AGENCY UNDERUTILIZED DISADVANTAGED BUSINESS</u> <u>ENTERPRISE UTILIZATION CERTIFICATION AND INSTRUCTIONS</u> (WSDOT FORM 272-056U)
- 7. <u>LOCAL AGENCY UNDERUTILIZED DISADVANTAGED BUSINESS ENTERPRISE</u> WRITTEN CONFIRMATION DOCUMENT (WSDOT FORM 422-031U)
- 8. < STATEMENT OF BIDDER'S QUALIFICATIONS
- 9. <u>CERTIFICATION OF COMPLIANCE WITH WAGE PAYMENT STATUTES</u>

After the Contract is awarded, execute:

- 1. LOCAL AGENCY CONTRACT (WSDOT FORM 272-006A)
- 2. LOCAL AGENCY PERFORMANCE BOND (WSDOT FORM 272-002)
- LOCAL AGENCY PAYMENT BOND (WSDOT FORM 272-003)

Failure to return this Declaration as part of the bid proposal package will make the bid nonresponsive and ineligible for award.

NON-COLLUSION DECLARATION

I, by signing the proposal, hereby declare, under penalty of perjury under the laws of the United States that the following statements are true and correct:

- That the undersigned person(s), firm, association or corporation has (have) not, either directly or indirectly, entered into any agreement, participated in any collusion, or otherwise taken any action in restraint of free competitive bidding in connection with the project for which this proposal is submitted.
- 2. That by signing the signature page of this proposal, I am deemed to have signed and to have agreed to the provisions of this declaration.

NOTICE TO ALL BIDDERS

To report rigging activities call:

1-800-424-9071

The U.S. Department of Transportation (USDOT) operates the above toll-free "hotline" Monday through Friday, 8:00 a.m. to 5:00 p.m., eastern time. Anyone with knowledge of possible bid rigging, bidder collusion, or other fraudulent activities should use the "hotline" to report such activities.

The "hotline" is part of USDOT's continuing effort to identify and investigate highway construction contract fraud and abuse and is operated under the direction of the USDOT Inspector General. All information will be treated confidentially and caller anonymity will be respected.

SR

DOT Form 272-0361 EF 07/2011

CITY OF FIRCREST CONTRACT DOCUMENTS KPG #19007 JUNE 2019

Proposal

Alameda Avenue Grind & Overlay Project Federal Aid No. STPUL-2991(003)

Note: Unit prices for all items, all extensions, and the total amount bid must be shown. Where conflict occurs between the unit price and the total amount specified for any item, the unit price shall prevail, and totals shall be corrected to conform thereto. Refer to Division 10 for definition of pay items. For SCHEDULE A. sales tax shall be included in the various unit and lump sum prices, per section 1-07.2(1) and WAC 458-20-171.

SCHEDULE A - ROADWAY IMPROVEMENTS

Item No.	Spec No.	Item Description	Qty	Unit	Unit Price	Amount
1	1-04	Minor Changes	1	EST	\$ 4,000.00	\$ 4,000.00
2	1-07	SPCC Plan	1	LS	\$720.00	\$720.00
3	1-09	Mobilization	1	LS	\$18,396.04	\$18,396.04
4	1-10	Project Temporary Traffic Control	1	LS	\$23,184.00	\$23,184.00
5	2-02	Adjust Water Valve to Finished Grade	4	EA	\$780.00	\$3,120.00
6	2-02	Adjust Sewer Manhole to Finished Grade	2	EA	\$1,000.00	\$2,000.00
7	2-02	Adjust Catch Basin to Finished Grade	1	EA	\$1,000.00	\$1,000.00
8	4-04	Crushed Surfacing Base Course	25	TON	\$6.00	\$150.00
9	5-04	Pavement Repair Excavation Incl. Haul	25	CY	\$24.99	\$624.75
10	5-04	Planing Bituminous Pavement	3380	SY	\$6.10	\$20,618.00
11	5-04	HMA Cl. 1/2 In. PG 64-22	420	TON	\$104.73	\$43,986.60
12	5-04	HMA Cl. 1/2 In. PG 64-22 for Pavement Repair	25	TON	\$104.73	\$2,618.25
13	7-05	Supply Sewer Manhole Ring and Cover	2	EA	\$442.20	\$884.40
14	8-01	Erosion/Water Pollution Control	1	LS	\$2,976.00	\$2,976.00
15	8-01	Inlet Protection	11	EA	\$50.00	\$550.00
16	8-20	Traffic Induction Loop	3	EA	\$1,800.00	\$5,400.00

CITY OF FIRCREST CONTRACT DOCUMENTS

Proposal

Alameda Avenue Grind & Overlay Project Federal Aid No. STPUL-2991(003)

Note: Unit prices for all items, all extensions, and the total amount bid must be shown. Where conflict occurs between the unit price and the total amount specified for any item, the unit price shall prevail, and totals shall be corrected to conform thereto. Refer to Division 10 for definition of pay items. For SCHEDULE A. sales tax shall be included in the various unit and lump sum prices, per section 1-07.2(1) and WAC 458-20-171.

SCHEDULE A - ROADWAY IMPROVEMENTS

Item No.	Spec No.	Item Description	Qty	Unit	Unit Price	Amount
17	8-22	Paint Line	1020	LF	\$6.00	\$6,120.00
18	8-22	Plastic Stop Line	15	LF	\$42.00	\$630.00
19	8-22	Plastic Traffic Letter	17	EA	\$106.80	\$1,815.60

Duildong Buchange

A F THE

Tan Dam waama

Alexand Handfor Carl

ditions Bourseast and rows how.

Local Agency Proposal -Signature Page

The bidder is hereby advised that by signature of this proposal he/she is deemed to have acknowledged all requirements and signed all certificates contained herein.

A proposal guaranty in an amount of five percent (5%) of the total bid, based upon the approximate estimate of quantities at the above prices and in the form as indicated below is attached hereto:

Cash	In the Amo	unt of				
Cashier's Check	□	Dollars				
Certified Check	□ (\$ ⊠) Payable to the State Treasurer				
Proposal Bond		In the Amount of 5% of the Bid				
Receipt is bereby ackn	owledged of addend	hum(s) No.(s)&&				
		Signature of Authorized Official(s)				
	Firm Name	Tony Lind Paving, LLC				
	Address	23048 172nd Ave SE				
		Kent, WA 98042				
State of Washington C	ontractor's License	NoTONYLLP922J9				
	Federal ID	No. 26-2186022				

Note:

- (1) This proposal form is not transferable and any alteration of the firm's name entered hereon without prior permission from the City of Fircrest will be cause for considering the proposal irregular and subsequent rejection of the bid.
- (2) Please refer to section 1-02.6 of the standard specifications, re: "Preparation of Proposal," or "Article 4" of the Instruction to Bidders for building construction jobs.

SR

DOT Form 272-036K EF 07/2011

CITY OF FIRCREST CONTRACT DOCUMENTS KPG #19007 JUNE 2019

LD-7

Local Agency Proposal Bond

KNOW ALL MEN BY THESE PRESENTS, That we, Tony Lind Paving LLC

of Kent, WA

as principal, and the Merchants National Bonding, Inc.

, and

a corporation duly organized under the laws of the state of lowa

authorized to do business in the State of Washington, as surety, are held and firmly bound unto the State of Washington in the full and penal sum of five (5) percent of the total amount of the bid proposal of said principal for the work hereinafter described, for the payment of which, well and truly to be made, we bind our heirs, executors, administrators and assigns, and successors and assigns, firmly by these presents.

The condition of this bond is such, that whereas the principal herein is herewith submitting his or its sealed proposal for the following highway construction, to wit:

ALAMEDA AVENUE GRIND & OVERLAY, FEDERAL AID NO. STPUL-2991 (003)

said bid and proposal, by reference thereto, being made a part hereof.

NOW, THEREFORE, If the said proposal bid by said principal be accepted, and the contract be awarded to said principal, and if said principal shall duly make and enter into and execute said contract and shall furnish bond as required by the *within a period of twenty (20) days from and after said award, exclusive of the day of such award, then this obligation shall be null and void, otherwise it shall remain and be in full force and effect. *City of Fircrest

IN TESTIMONY WHEREOF, The principal and surety have caused these presents to be signed

and sealed this _____ 28th _____ day of _____ 2019

Tony Lind Paving LLC (Principal)

Merchants National Bonding, Inca

(Surety)

(Attomey-in-fact) Heather L. Allen

DOT Form 272-001A EF 07/2011

SR

CITY OF FIRCREST CONTRACT DOCUMENTS KPG #19007 JUNE 2019



Know All Persons By These Presents, that MERCHANTS BONDING COMPANY (MUTUAL) and MERCHANTS NATIONAL BONDING, INC., both being corporations of the State of Iowa (herein collectively called the "Companies") do hereby make, constitute and appoint, individually,

Aliceon A Keltner; Annelies M Richie; Brandon K Bush; Brent E Heilesen; Carley Espiritu; Christopher Kinyon; Cynthia L Jay; Eric A Zimmerman; Erica E Mosley; Heather L Allen; Holli Albers; James B Binder; Jamie L Margues; Karen C Swanson; Kyle Joseph Howat; Tamara A Ringeisen

their true and lawful Attorney(s)-in-Fact, to sign its name as surety(ies) and to execute, seal and acknowledge any and all bonds, undertakings, contracts and other written instruments in the nature thereof, on behalf of the Companies in their business of guaranteeing the fidelity of persons, guaranteeing the performance of contracts and executing or guaranteeing bonds and undertakings required or permitted in any actions or proceedings allowed by law.

This Power-of-Attorney is granted and is signed and sealed by facsimile under and by authority of the following By-Laws adopted by the Board of Directors of Merchants Bonding Company (Mutual) on April 23, 2011 and amended August 14, 2015 and adopted by the Board of Directors of MerchantsNational Bonding, Inc., on October 16, 2015.

"The President, Secretary, Treasurer, or any Assistant Treasurer or any Assistant Secretary or any Vice President shall have power and authority to appoint Attorneys-in-Fact, and to authorize them to execute on behalf of the Company, and attach the seal of the Company thereto, bonds and undertakings, recognizances, contracts of indemnity and other writings obligatory in the nature thereof."

"The signature of any authorized officer and the seal of the Company may be affixed by facsimile or electronic transmission to any Power of Attorney or Certification thereof authorizing the execution and delivery of any bond, undertaking, recognizance, or other suretyship obligations of the Company, and such signature and seal when so used shall have the same force and effect as though manually fixed."

In connection with obligations in favor of the Florida Department of Transportation only, it is agreed that the power and aut hority hereby given to the Attorney-in-Fact includes any and all consents for the release of retained percentages and/or final estimates on engineering and construction contracts required by the State of Florida Department of Transportation. It is fully understood that consenting to the State of Florida Department of Transportation and/or its assignee, shall not relieve this surety company of any of its obligations under its bond.

In connection with obligations in favor of the Kentucky Department of Highways only, it is agreed that the power and authority hereby given to the Attorney-in-Fact cannot be modified or revoked unless prior written personal notice of such intent has been given to the Commissioner-Department of Highways of the Commonwealth of Kentucky at least thirty (30) days prior to the modification or revocation.

In Witness Whereof, the Companies have caused this instrument to be signed and sealed this 20th day of May , 2019



MERCHANTS BONDING COMPANY (MUTUAL) MERCHANTS NATIONAL BONDING, INC.

President

STATE OF IOWA

COUNTY OF DALLAS ss.

On this tis 20th day of May 2019, before me appeared Larry Taylor, to me personally known, who being by me duly sworn did say that he is President of MERCHANTS BONDING COMPANY (MUTUAL) and MERCHANTS NATIONAL BONDING, INC.; and that the seals affixed to the foregoing instrument are the Corporate Seals of the Companies; and that the said instrument was signed and sealed in behalf of the Companies by authority of their respective Boards of Directors.



Notary Public

(Expiration of notary's commission does not invalidate this instrument)

I, William Warner, Jr., Secretary of MERCHANTS BONDING COMPANY (MUTUAL) and MERCHANTS NATIONAL BONDING, INC., do hereby certify that the above and foregoing is a true and correct copy of the POWER-OF-ATTORNEY executed by said Companies, which is still in full force and effect and has not been amended or revoked.

In Witness Whereof, I have hereunto set my hand and affixed the seal of the Companies on this 28th day of June , 2019 .



Local Agency Certification for Federal-Aid Contracts

The prospective participant certifies by signing and submitting this bid or proposal, to the best of his or her knowledge and belief, that:

(1) No Federal appropriated funds have been paid or will be paid, by or on behalf of the undersigned, to any person for influencing or attempting to influence an officer or employee of any Federal agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with the awarding of any Federal contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment, or modification of any Federal contract, grant, loan or cooperative agreement.

(2) If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any Federal agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with this Federal contract, grant, loan, or cooperative agreement, the undersigned shall complete and submit Standard Form-LLL, "Disclosure Form to Report Lobbying," in accordance with its instructions.

This certification is material representation of the fact upon which reliance was placed when this transaction was made or entered into. <u>Submission of this certification is a prerequisite for making or entering into this transaction</u> imposed by Section 1352, Title 31, U.S. Code. Any person who fails to file the required certification shall be subject to a civil penalty of not less than \$10,000 and not more than \$10,000 for each failure.

The prospective participant also agrees by submitting his or her bid or proposal that he or she shall require that the language of this certification be included in all lower tier subcontracts, which exceed \$100,000 and that all such subrecipients shall certify and disclose accordingly.

DOT Form 272-040A EF 07/2011

SR

CITY OF FIRCREST CONTRACT DOCUMENTS KPG #19007 JUNE 2019

LD-9



Underutilized Disadvantaged Business **Enterprise Utilization Certification**

To be eligible for Award of this Contract the Bidder shall fill out and submit, as a supplement to its sealed Bid Proposal, an Underutilized Disadvantaged Business Enterprise (UDBE) Utilization Certification. The Contracting Agency shall consider as non-responsive and shall reject any Bid Proposal that does not contain a UDBE Utilization Certification which properly demonstrates that the Bidder will meet the UDBE participation requirements in one of the manners provided for in the proposed Contract. Refer to the instructions on Page 2 when filling out this form or the Bid may be rejected. An example form has been provided on Page 3. The successful Bidder's UDBE Utilization Certification shall be deemed a part of the resulting Contract.

Box 1: **Tony Lind Paving**, **LLC** cortifies that the UDBE firms listed below have been contacted regarding participation on this project. If this Bidder is successful on this project and is awarded the Contract, it shall assure that subcontracts or supply agreements are executed with named UDBEs. (If necessary, use additional sheets.)

Box 2: ALAMEDA AVENUE GRIND & OVERLAY

Column 1	Column 2	Column 3	Column 4	Column 5
Name of UDBE (See Instructions)	Project Role (See instructions)	Description of Work (See instructions)	Dollar Amount Subcontracted to UDBE (See instructions)	Dollar Amount to be Applied Towards Goal (See instructions)
Advanced Government Services, Inc.	Subcontractor	Traffic Control	\$19,000.00	\$7,000.00
G & G Incorporated	Subcontractor	Electrical	\$4,500.00	\$4,500.00
0				

Inderutilized Disadvantaged Business Enterprise Condition of Award Contract Goal

\$11,103.49 Total UDBE Commitment Dollar Amount \$11,500.00

Box 4

5 By checking Box 5 the Bidder is stating that their attempts to solicit sufficient UDBE participation to meet the COA Contract goal has been unsuccessful and good faith effort will be submitted in accordance with Section 1-02.9 of the Contract

Box 3

1001 Form 272-056U Revised 02/2018

CITY OF FIRCREST CONTRACT DOCUMENTS

KPG #19007 JUNE 2019

LD-10



Underutilized Disadvantaged Business Enterprise (UDBE) Written Confirmation Document

See Contract Provisions: UDBE Document Submittal Requirements Disadvantaged Business Enterprise Participation

THIS FORM SHALL ONLY BE SUBMITTED TO A UDBE THAT IS LISTED ON THE CONTRACTOR'S UNDERUTILIZED DISADVANTAGED BUSINESS ENTERPRISE UTILIZATION CERTIFICATION.

THE CONTRACTOR SHALL COMPLETE PART A PRIOR TO SENDING TO THE UDBE.

PART A: To be completed by the bidder

The entries below shall be consistent with what is shown on the Bidder's Underutilized Disadvantaged Business Enterprise Utilization Certification. Failure to do so will result in Bid rejection.

Contract Title:	Alameda Ave Grind and Overlay
Bidder's Business Name:	Tony Lind Paving, LLC
UDBE's Business Name:	Advance Government Services
Description of UDBE's W	Vork: Traffic Control
Dollar Amount to be App	lied Towards UDBE Goal: \$7,000
Dollar Amount to be Subo	*Optional Field

PART B: To be completed by the Underutilized Disadvantaged Business Enterprise

As an authorized representative of the Underutilized Disadvantaged Business Enterprise, I confirm that we have been contacted by the Bidder with regard to the referenced project for the purpose of performing the Work described above. If the Bidder is awarded the Contract, we will enter into an agreement with the Bidder to participate in the project consistent with the information provided in Part A of this form.

Name (printed):	Arti O'Brien			
Signature:	1ATO Burn		1	
Title:	President and Owner	×	93 	
Address:	8644 Pacific Avenue,	Date: _	6/27/2019	
	Tacoma WA, 98444			



See Contract Provisions: UDBE Document Submittal Requirements Disadvantaged Business Enterprise Participation

THIS FORM SHALL ONLY BE SUBMITTED TO A UDBE THAT IS LISTED ON THE CONTRACTOR'S UNDERUTILIZED DISADVANTAGED BUSINESS ENTERPRISE UTILIZATION CERTIFICATION.

THE CONTRACTOR SHALL COMPLETE PART A PRIOR TO SENDING TO THE UDBE.

PART A: To be completed by the bidder

The entries below shall be consistent with what is shown on the Bidder's Underutilized Disadvantaged Business Enterprise Utilization Certification. Failure to do so will result in Bid rejection.

Contract Title:	alameda AUE Grindi	FOUErlay
	: Tony Lind Paving, LLC	
UDBE's Business Name:	GIGI THE	
Description of UDBE's W	Vork: Electrical	
Dollar Amount to be App	lied Towards UDBE Goal: 🎂 🐴 🤧	
Dollar Amount to be Subc	Contracted to UDBE*:	

PART B: To be completed by the Underutilized Disadvantaged Business Enterprise

As an authorized representative of the Underutilized Disadvantaged Business Enterprise, I confirm that we have been contacted by the Bidder with regard to the referenced project for the purpose of performing the Work described above. If the Bidder is awarded the Contract, we will enter into an agreement with the Bidder to participate in the project consistent with the information provided in Part A of this form.

Name (printed):	Bonnie Grahaw	n
Signature:	63	
Title:	President	
Address:	KENT, WA 98042	Date: 6282019

DOT Form 422-031U

CITY OF FIRCREST STATEMENT OF BIDDER'S QUALIFICATIONS

Contractor Name:	Tony Lind Paving, LLC	Contact: Kevin Christoperson
Business Address:	23048 172nd Ave SE Kent,	WA 98042
Business phone:	253-630-7612	Fax:

Number of years the Contractor has been engaged in the construction business under the present firm name: 22

Describe the general character of work performed by your company: <u>Grading</u>, <u>HMA</u> paving Concrete, General construction

List five projects of a similar nature which Contractor has completed within the last 10 years. Include contract amount and contact information for references:

Project Name	Amount	Owner/Agency	Contact	Phone	Year Completed
JT-279508	\$19,500.00	Kirkland	Ralph Utley	360.739.3679	2019
E.743972	\$94,777.00	Fife	Haley McLaughlin	206.347.8776	2017
Boeing DC-9	\$1,379,000.00	Boeing	Chet Williams	206.713.0984	2018
Boeing DC-9	\$1,101,000.00	Boeing	Chet Williams	206.713.0984	2018
Thompson Valley Rd.	\$179,000.00	Sumner	Andrew Leach	253.299.5711	2017

See the supplemental bidder qualification requirements specified in Special Provision Section 1-02.1.

List major equipment anticipated to be used on this project; indicate whether Contractor-owned or to be leased from others:

HMA Pavers (owned + Leesed) Steel wheel Rollers (owned a leased) En my trucks lowned

Bank reference(s):

Washington State Contractor Registration No.: _________

Uniform Business Identification No.: 602-813-830

I certify that other contracts now in progress or hereafter obtained will not interfere with timely performance of the City of Fircrest project should I become the successful bidder.

Authorized Signature:

Print Name: Tony Lind

Title: Owner, Member Manager

CITY OF FIRCREST CONTRACT DOCUMENTS

Certification of Compliance with Wage Payment Statutes

The bidder hereby certifies that, within the three-year period immediately preceding the bid solicitation date (**June 7th, 2019**), that the bidder is not a "willful" violator, as defined in RCW 49.48.082, of any provision of chapters 49.46, 49.48, or 49.52 RCW, as determined by a final and binding citation and notice of assessment issued by the Department of Labor and Industries or through a civil judgment entered by a court of limited or general jurisdiction.

I certify under penalty of perjury under the laws of the State of Washington that the foregoing is true and correct.

Tony Lind Pa	ving, LLC	
Contractor	\mathcal{O}	
Signature 6f Author	rized Official*	
Tony Lind		_
Printed Name		-
Owner, Membe	r Manager	_
Title		
06/28/2019	Kent	WA
Date	City	State
Check one:		
Individual	Partnership Jo	int Venture X Corporation
State of Incorporation	on, or if not a corporation, St	ate where business entity was formed:

If a Co-partnership, give firm name under which business is transacted:

* If a corporation, proposal must be executed in the corporate name by the president or vice-president (or any other corporate officer accompanied by evidence of authority to sign). If a co-partnership, proposal must be executed by a partner.

CITY OF FIRCREST CONTRACT DOCUMENTS

KPG #19007 JUNE 2019

LD-14

PRIME CONTRACTOR

Tony Lind Paving

Home	Español	Contact	Search	L&I		SEARCH
			A-Z	Index	Help	My L&I
Safety & Health	Claim	s & Insurance	Workplace R	ights	Tra	des & Licensing

23048 172ND AVE SE KENT, WA 98042

Limited Liability Company

Meets current requirements.

253-630-7612

KING County

Business type



TONY LIND PAVING LLC

Owner or tradesperson Principals LIND, ANTHONY AESOPH, PARTNER/MEMBER Doing business as TONY LIND PAVING LLC

WA UBI No...

Certifications & Endorsements

OMWBE Certifications No active certifications exist for this business.

Apprentice Training Agent

Not allowed to have apprentices.

License

Verify the contractor's active registration / license / certification (depending on trade) and any past violations.

License specialties **GENERAL** License no. **TONYLLP922J9** Effective — expiration 04/29/2008— 05/01/2020

Construction Contractor

Bond

Developers Surety & Indem Co Bond account no. 674040C

Received by L&I 11/01/2017

Bond history

Insurance Union Insurance Company Policy no. CPA603236121

Received by L&I 12/21/2018

\$12,000.00

Active.

Effective date 11/19/2017 Expiration date Until Canceled

\$1,000,000.00

Effective date 01/01/2019 Expiration date 01/01/2020

Insurance history

Savings

No savings accounts during the previous 6 year period.

Lawsuits against the bond or savings No lawsuits against the bond or savings accounts during the previous 6 year period.

L&I Tax debts No L&I tax debts are recorded for this contractor license during the previous 6 year period, but some debts may be recorded by other agencies.

License Violations No license violations during the previous 6 year period.

Workers' comp

Do you know if the business has employees? If so, verify the business is up-to-date on workers' comp premiums,

L&I Account ID 579,346-02 Account is current.

Doing business as TONY LIND PAVING Estimated workers reported Quarter 1 of Year 2019 "11 to 20 Workers" L&I account contact T3 / DAVID PRYOR (360)902-4804 - Email: PRYD235@Ini.wa.gov

Public Works Requirements

Verify the contractor is eligible to perform work on public works projects.

Required Training– Effective July 1, 2019 Exempt from this requirement.

Contractor Strikes No strikes have been issued against this contractor.

Contractors not allowed to bid No debarments have been issued against this contractor.

Workplace safety and health

No inspections during the previous 6 year period.

© Washington State Dept. of Labor & Industries. Use of this site is subject to the laws of the state of Washington.

	Home	Español	Contact	Search L&I		SEARCH	
				A-Z Index	Help	My L&I	
	Safety & Health	Claim	is & Insurance	Workplace Rights	Trade	s & Licensing	
Washington State Department of Labor & Industries							
Contractor Strike List							
This is a list of contractors who have one or mor registration law, industrial insurance law, or ap options presented below.							

Company Name: Tony Lind Paving, LLC	Principal:	
WA UBI Number: 602 813 830	License Number:	TONYLLP922J9
RCW: All		
From: MM/DD/YYYY To: MM/DD/YYYY		
Apply Filters Reset		
		Download all strike data 🚯
Show 25 🗸 per page		
Showing 0 records		
FirstPreviousNextLast		
Company Name UBI License Principals RCW Effective Da	ite	
There are no records that match your search criteria		
Show 25 🗸 per page		
Showing 0 records		
FirstPreviousNextLast		

© Washington State Dept, of Labor & Industries. Use of this site is subject to the laws of the state of Washington.

	Home	Español	Contact	Search L&I		SEARCH
				A-Z Index	Help	My L&I
Safety & He	alth	Claims	s & Insurance	Workplace Rights	Trades	& Licensing



Debarred Contractors List

A debarred contractor may not bid on, or have a bid considered on, any public works contract. You can search and filter this list using the options presented below.

Company Name Tony Lind Paving, LLC WA UBI Number 602 813 830 License Number Principal [RCW: All V From MM/DD/YYYY To: 06/28/2019 Penalty Due: All 🗸 Wage Due: All V Apply Filters Reset Download all debarment data Show 25 🗸 per page Showing 0 records FirstPreviousNextLast Company Name UBI License Principals Status RCW Debar Begins Debar Ends Penalty Due Wages Due There are no records that match your search criteria. Show 25 🗸 per page Showing 0 records **FirstPreviousNextLast**

© Washington State Dept. of Labor & Industries. Use of this site is subject to the laws of the state of Washington.

Washington State Department of Revenue

Services B	usiness Lookup	TONY LIND PAVING LLC		
Tax Information			New search	ch Back to results
Entity name:	TONY LIND PAV	ING LLC		
Entity type:	Limited Liability C	Company		
Excise tax account ID #:	602-813-830			
UBI #:	602-813-830			
Opened:	April 1, 2008			
Closed:				
Mailing address:	23048 172ND AV KENT WA 98042			
NAICS:	237310 - Highway	v, Street, and Bridge Construction		

Reseller Permit(s)

Reseller Permit(s)				Filter	
Reseller permit #	Status		Effective date	Expiration date	
A16765719	Active		Jan-01-2018	Dec-31-2019	
A16765717	Expired		Jan-01-2016	Dec-31-2017	
A16765715	Expired		Jan-01-2014	Dec-31-2015	
A16765713	Expired		Jan-01-2012	Dec-31-2013	
Business License Loc	ations			Filter	
Business name		License account ID #	Location address		
TONY LIND PAVING		602813830-001-0001	23048 172ND AVE SE KENT WA 98042-4713		

The Business Lookup information is updated nightly. Search date and time: 6/28/2019 12:13:16 PM

Working together to fund Washington's future

SUBCONTRACTORS

Home	Español	Contact	Search	Search L&I		SEARCH
			A-Z	Index	Help	My L&I
Safety & Health	Claims	s & Insurance	Workplace R	ights	Trac	des & Licensing



ADVANCED GOVERNMENT SERV INC

Owner or tradesperson Principals O'Brien, John R, MEMBER

O'Brien, Arti D, MEMBER

8644 Pacific Ave **TACOMA, WA 98444** 253-531-9782 **PIERCE County**

Doing business as ADVANCED GOVERNMENT SERV INC

WA UBI No. 603 585 454

Business type Corporation Governing persons ARTI D OBRIEN JOHN R OBRIEN;

Certifications & Endorsements

OMWBE Certifications

Disadvantaged Business Enterprise (DBE) Minority/Women Business Enterprise (MWBE)

Apprentice Training Agent Not allowed to have apprentices.

License

Verify the contractor's active registration / license / certification (depending on trade) and any past violations. **Construction Contractor**

Active. Meets current requirements.

License specialties GENERAL	
License no. ADVANGS834R9	
Effective — expiration 12/29/2017— 12/29/2019	
Bond	
Western Surety Co	\$12,000.00
Bond account no. 63026730	
Received by L&I 12/29/2017	Effective date 06/02/2017
	Expiration date Until Canceled
Insurance	
WESTERN NATIONAL MUTUAL INS	\$1,000,000,00

WESTERN NATIONAL MUTUAL INS Policy no.

\$1,000,000.00

CCP1157687-02

Received by L&I 12/31/2018 Effective date 01/01/2019 Expiration date 01/01/2020

Insurance history

Savings

No savings accounts during the previous 6 year period.

Lawsuits against the bond or savings

No lawsuits against the bond or savings accounts during the previous 6 year period.

L&I Tax debts

No L&I tax debts are recorded for this contractor license during the previous 6 year period, but some debts may be recorded by other agencies.

License Violations

No license violations during the previous 6 year period.

Workers' comp

Do you know if the business has employees? If so, verify the business is up-to-date on workers' comp premiums,

L&I Account ID 254,819-01

Account is current.

Doing business as ADVANCED GOVERNMENT SERVICES

Estimated workers reported Quarter 1 of Year 2019 "31 to 50 Workers" L&I account contact

T4 / STEPHEN TASSONI (360)902-4819 - Email: TASS235@Ini.wa.gov

Public Works Requirements

Verify the contractor is eligible to perform work on public works projects.

Required Training– Effective July 1, 2019 Exempt from this requirement. Completed the training on 1/18/2019

Contractor Strikes No strikes have been issued against this contractor.

Contractors not allowed to bid No debarments have been issued against this contractor.

Workplace safety and health

No inspections during the previous 6 year period.

© Washington State Dept. of Labor & Industries. Use of this site is subject to the laws of the state of Washington.

<u>Print</u>

Business & Contact Information						
BUSINESS NAME Advanced Government Services Inc, DBA Advanced Government Services Inc						
OWNER	Arti O'Brien					
ADDRESS	8644 PACIFIC AVENUE Tacoma, WA 98444	Map This Address				
PHONE	206-455-4980					
FAX	425-427-6332					
EMAIL	arti@advancedgovernmentservices.com					
COUNTY	Pierce (WA)					

Certification Inform	ation
CERTIFYING AGENCY	Washington State Office of Minority & Women's Business Enterprises
CERTIFICATION TYPE	MWBE - Minority/Women Business Enterprise
CERTIFIED BUSINESS DESCRIPTION	Full service Flagging and traffic control services including the installation and removal of Class A and Class B Signs. Computer generated traffic plans. Rental and sales of new and used traffic control equipment including: NCHRP 350 crash worthy signs and stands, arrow boards, trailer & trucks, PCM boards, light towers, traffic control vehicles with flashing beacon, pilot cars, and TMA. In store and online retail sales of safety apparel including gloves, safety glasses, hard hats, vests/jackets, raingear, parkas and light wands.

Commodity Codes				
Code	Description			
NAICS 561990	Flagging (i.e., traffic control) services			
NAICS 237310	Sign erection, highway, road, street, or bridge			
NAICS 448140	Unisex clothing stores			
NAICS 452319	Variety stores			
NAICS 453310	Used merchandise stores			

NAICS 532310 General rental centers

1	
603585454	
M4F0024875	
r	

Home	Español	Contact	Search L&I		SEARCH
			A-Z Index	Help	My L&I
Safety & Health	Claims	s & Insurance	Workplace Rights	Trades	& Licensing



Debarred Contractors List

A debarred contractor may not bid on, or have a bid considered on, any public works contract. You can search and filter this list using the options presented below.

Company Name WA UBI Number: 603 585 454 License Number Principal RCW: All V From MM/DD/YYYY To: MM/DD/YYYY Penalty Due: All 🗸 Wage Due: Apply Filters Reset Download all debarment data Show 25 🗸 per page Showing 0 records FirstPreviousNextLast Company Name UBI License Principals Status RCW Debar Begins Debar Ends Penalty Due Wages Due There are no records that match your search criteria, Show 25 🗸 per page Showing 0 records **FirstPreviousNextLast**

© Washington State Dept. of Labor & Industries. Use of this site is subject to the laws of the state of Washington

	Home	Español	Contact		Search L&I		SEARCH
					A-Z Index	l (elp	My L&I
Safety & He	ealth	Claim	s & Insurance	Workp	lace Rights	Trade	s & Licensing



Contractor Strike List

This is a list of contractors who have one or more strikes (violations) under prevailing wage law, contractor registration law, industrial insurance law, or apprenticeship law. You can search and filter this list using the options presented below.

Company Name:		Principal:	
WA UBI Number:	603 585 454	License Number:	
RCW: All	\checkmark		
From: MM/DD/YY	YY To: MM/DD/YYYY		
	Apply Filters Reset		Download all strike data 🚯
Show 25 v per pag Showing 0 records FirstPreviousNextLas			
Company Name UBI L	icense Principals RCW Effective Da	te	
Show 25 v per pag Showing 0 records			
FirstPreviousNextLas	t		

© Washington State Dept. of Labor & Industries. Use of this site is subject to the laws of the state of Washington

	Home Español Contact	Search L&I	SEARCH
		A-Z Index	c Help My L&I
	Safety & Health Claims & Insur	rance Workplace Rights	Trades & Licensing
Washington State Department of			
Labor & Industries			
Owner or tradesperson Principals	18044 SE 224TH ST KENT, WA 98042 425-432-1325		
GRAHAM, BONNIE J, PRESIDENT McDowell, Kamie, VICE PRESIDENT	KING County		
GRAHAM, BONNIE, AGENT			
TENORIO, SIRENA T			
(End: 08/01/2014) GERARD, CHARLES			
(End: 08/01/2014)			
GRAHAM, PAUL D			
(End: 08/01/2014)			
Doing business as			
WA UBI No. 601 381 635	Business type Corporation		
^D arent company 3 & G INCORPORATED			
Certifications & Endorsem	ents		
MWBE Certifications			
sadvantaged Business Enterprise (DBE)			
sadvantaged Business Enterprise (DBE)			
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE)			
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE)			
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) oprentice Training Agent of allowed to have apprentices.			
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) oprentice Training Agent at allowed to have apprentices.	ense / certification (depending on trade) and ar	1y past violations.	
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) oprentice Training Agent at allowed to have apprentices.	ense / certification (depending on trade) and ar Active. Meets current requireme		
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) operatice Training Agent of allowed to have apprentices. License rify the contractor's active registration / lic Electrical Contractor	Active.		ĸ
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) oprentice Training Agent of allowed to have apprentices. License erify the contractor's active registration / lic Electrical Contractor License specialties GENERAL License no.	Active.		۰.
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) oprentice Training Agent of allowed to have apprentices. License wify the contractor's active registration / lice Electrical Contractor License specialties SENERAL License no. GGCOR**081MU	Active.		κ.
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) oprentice Training Agent of allowed to have apprentices. License rify the contractor's active registration / lic Electrical Contractor	Active.		` .
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) operentice Training Agent of allowed to have apprentices. License rify the contractor's active registration / lic Electrical Contractor License specialties SENERAL License no. GGCOR**081MU Effective — expiration 17/31/1992— 08/02/2020 Designated administrator	Active.	ents.	
sadvantaged Business Enterprise (DBE) omen Business Enterprise (WBE) oprentice Training Agent of allowed to have apprentices. License wify the contractor's active registration / lic Electrical Contractor License specialties SENERAL License no. GGCOR**081MU Effective — expiration	Active. Meets current requireme Active.	ents.	•

https://saaura lai wa aau/warifu/Datail aanv911DI-601201625 &I IC-CCCOD**001NAII& CAW-

No bond accounts during the previous 6 year period.

Savings

(in lieu of bond) Received by L&I 12/22/2004 Release date N/A Savings account ID 253553002630

(in lieu of bond) Received by L&I 10/26/2004 Release date N/A Savings account ID 6033431628

(in lieu of bond) Received by L&I

Release date N/A Savings account ID 0220026694

License Violations

Infraction no. ESHAU01006 Issue date 09/22/2015 Violation city KENT Type of violation ELECTRICAL CITATION Description

Failing to request an inspection within 3 business days after completion of the installation or 1 business day after energizing, whichever occurs first as referenced in WAC 296-46B-901. Failing to request an inspection for EWPS- 2402784E, 2402802E, 2402809E, 2402814E, 2402819E & 2402831E.

Infraction no. EMIND00257 Issue date 08/05/2014 Violation city Kent Type of violation ELECTRICAL CITATION Description Employing an individual for the purposes of chapter 19.28 RCW who does not possess a valid certificate of competency or training

Violation amount **\$250.00**

Workers' comp

certificate to do electrical work.

Do you know if the business has employees? If so, verify the business is up-to-date on workers' comp premiums.

L&I Account ID 823,490-00 Account is current.

Help us improve

\$4,000.00 Effective date **09/25/1997** Impaired date **11/01/2004**

\$4,000.00 Effective date 07/31/1992 Impaired date N/A

Satisfied RCW/WAC 19.28.101 RCW Violation amount \$250.00

Satisfied

RCW/WAC

19.28.271 RCW

Public Works Requirements

Verify the contractor is eligible to perform work on public works projects.

Required Training– Effective July 1, 2019 Exempt from this requirement.

Contractor Strikes No strikes have been issued against this contractor.

Contractors not allowed to bid No debarments have been issued against this contractor.

Workplace safety and health

Check for any past safety and health violations found on jobsites this business was responsible for.

Inspection results date 09/16/2014 Inspection no. 317401362 Location 120th Ave and NE 6th Street Bellevue, WA 98005

Inspection results date 09/10/2013 Inspection no. 316868058 Location 120th Ave NE and NE 6th St Bellevue, WA 98005

Violations

Violations

© Washington State Dept, of Labor & Industries. Use of this site is subject to the laws of the state of Washington.

flome	Español	Contact	Search L&I		SEARCH
			A-Z Index	Help	My L&I
Safety & Health	Clain	ns & Insurance	Workplace Rights	Tra	des & Licensing



Debarred Contractors List

A debarred contractor may not bid on, or have a bid considered on, any public works contract. You can search and filter this list using the options presented below.

© Washington State Dept. of Labor & Industries. Use of this site is subject to the laws of the state of Washington

Home	Español Contact	Search L&I	SEARCH
		A-Z Index	Help My L&I
Safety & Health	Claims & Insurance	Workplace Rights	Trades & Licensing



Contractor Strike List

This is a list of contractors who have one or more strikes (violations) under prevailing wage law, contractor registration law, industrial insurance law, or apprenticeship law. You can search and filter this list using the options presented below.

Company Name: Principal:	
WA UBI Number: 601 381 635 License Number	
RCW: All	
From: MM/DD/YYYY To: MM/DD/YYYY	
Apply Filters Reset	
	Download all strike data 🕢
Show 25 🗸 per page	
Showing 0 records	
FirstPreviousNextLast	
Company Name UBI License Principals RCW Effective Date	
There are no records that match your search criteria.	
Show 25 🗸 per page	
Showing 0 records	
FirstPreviousNextLast	

© Washington State Dept. of Labor & Industries. Use of this site is subject to the laws of the state of Washington

CLOSE WINDOW	×

Print

Business & Contact Information				
BUSINESS NAME	G & G Incorporated, DBA G & G Incorporate	d		
OWNER	Ms. Bonnie Graham			
ADDRESS	18044 SE 224TH ST Ma Kent, WA 98042	p This Address		
PHONE	425-432-1325			
FAX	425-432-8155			
EMAIL	gginc1992@aol.com			
COUNTY	King (WA)			

Certification Information						
CERTIFYING AGENCY	Washington State Office of Minority & Women's Business Enterprises					
CERTIFICATION TYPE	DBE - Disadvantaged Business Enterprise					
CERTIFIED BUSINESS DESCRIPTION	Electrical contracting, traffic signals, loops, street lighting, road sign erection, pole removal and installation, concrete cutting for traffic signal work.					

Commodity Codes

Code	Description
NAICS 237310	Concrete paving (i.e., highway, road, street, public sidewalk)
NAICS 237310	Sign erection, highway, road, street, or bridge
NAICS 238210	Communication equipment installation
NAICS 238210	Highway, street and bridge lighting and electrical signal installation
NAICS 238990	Pole (e.g., telephone) removal

Additional Information

1	UDBE	Yes	
	SBE CERTIFICATION	Yes	
	UBI #	601381635	
	CERTIFICATION NUMBER	D5F0010391	
I			

City of Fircrest

Alameda Ave Grind & Overlay Project STPUP-2991(003)

Reference Checks for Tony Lind Paving, LLC

1st Reference:

City of Kirkland Name of reference: Ralph Utley "They do a lot of our grind and overlays and we have always been very pleased with their work. Paperwork has never been an issue for them"

2nd Reference:

Boeing

Name of reference: Chet Williams "We have worked with them on a lot of Boeing work and with municipalities and they do excellent work and we have never had a problem with them in any way."

3rd Reference:

City of Sumner Name of reference: Andrew Leach

"They worked for us a couple years ago and I remember they were pretty good and we had no problems. We even had some adds on that project and they were good to work with. The project was not Federally Funded."

8% UDBE Goal % Apprentice Goal – N/A

References checked by Daniel Clark, KPG

NEW BUSINESS: KPG, P.S. Supplement #1 for Construction Management Services ITEM 10B.

FROM: Jerry Wakefield, Public Works Director

RECOMMENDED MOTION: I move to adopt Resolution No. _____, authorizing the City Manager to execute Supplement #1 to the Local Agency A&E Professional Services Negotiated Hourly Rate Agreement with KPG, P.S. for providing construction management services for the Alameda Avenue Grind and Overlay project.

PROPOSAL: The Council is being asked to authorize the City Manager to enter into supplement #1 to the Local Agency A&E Professional Services Negotiated Hourly Rate Agreement with KPG, P.S. for providing construction management services for the Alameda Avenue Grind and Overlay project. This contract has been reviewed and approved by WSDOT as it is required to be per their contract requirements.

FISCAL IMPACT: The agreement amount will result in a not-to-exceed budget of \$37,787.00. This cost is included in the grant monies for this project and has been obligated through WSDOT. There is a 13.5% match monies needed from the City for this phase of the project, which is \$5,096.25 and is in this year's budget amount. The obligation from WSDOT includes \$178,000.00 for construction. This includes construction cost of \$138,793.64 and \$37,787.00 = \$176,580.64.

ADVANTAGE: Utilizing KPG, P.S. for this phase of work and as authorized by WSDOT will provide the City the support needed during the construction phase of this project.

DISADVANTAGES: None identified.

ALTERNATIVES: Not to proceed with the design and bid call services of the project at this time and return the grant.

HISTORY: In March 2019, the City Council approved the Local Agency A&E Professional Services Negotiated Hourly Rate Agreement with KPG, P.S. for the design phase of this project. As this project now has been authorized for construction, a supplement to the agreement is needed to retain KPG, P.S. for the construction management services. WSDOT has approved this supplement along with the award of the construction contract for this project.

ATTACHMENTS: <u>Resolution</u> <u>Supplement #1 Agreement</u>

1	
2	CITY OF FIRCREST RESOLUTION NO
3	A RESOLUTION OF THE CITY COUNCIL OF THE CITY OF
4	FIRCREST, WASHINGTON, AUTHORIZING THE CITY MANAGER TO EXECUTE SUPPLEMENT #1 TO THE LOCAL AGENCY A&E
5	PROFESSIONAL SERVICES NEGOTIATED HOURLY RATE
6	AGREEMENT WITH KPG, P.S. FOR PROVIDING CONSTRUCTION MANAGEMENT SERVICES FOR THE ALAMEDA AVENUE GRIND
7	AND OVERLAY PROJECT.
8 9	WHEREAS , the City of Fircrest had conducted a selection process and selected KPG, P.S. who prepared the design and bid call services for the Alameda Avenue Grind and Overlay project; and
10 11	WHEREAS, the City of Fircrest has secured the necessary funding to finance the work identified in the Alameda Avenue Grind and Overlay project; and
12	WHEREAS, the City wishes to contract with KPG, P.S., to provide the construction management services for the Alameda Avenue Grind and Overlay project. Now, Therefore,
13	BE IT RESOLVED BY THE CITY COUNCIL OF THE CITY OF FIRCREST THAT:
14	
15 16	Section 1. The City Manager is hereby authorized and directed to execute supplement #1 to the Local Agency A&E Professional Services Negotiated Hourly Rate Agreement with KPG, P.S. for providing construction management services for the Alameda Avenue Grind and Overlay project in an amount not to exceed \$37,787.00.
17	APPROVED AND ADOPTED BY THE CITY COUNCIL OF THE CITY OF
18	FIRCREST, WASHINGTON, at a regular meeting thereof this 9th day of July 2019.
19	APPROVED:
20	
21	Denny Waltier, Mayor Pro Tempore
22	ATTEST:
23	
24 25	Jessica Nappi, City Clerk
25 26	Jessica Nappi, City Clerk
26 27	APPROVED AS TO FORM:
27	
28 29	Michael B. Smith, City Attorney
2) 30	The more B. Shindi, City Attorney
31	Page 1 of 1
32	

Washington State Department of Transportation

Supplemental Agreement Number _1	Organization and Address KPG, P.S. 2502 Jefferson Avenue	
Original Agreement Number STPUL-2991(003)	Tacoma, WA 98402 Phone: 253-627-0720	
Project Number	Execution Date	Completion Date
KPG #19007		December 31, 2020
Project Title	New Maximum Amount Payable	
Alameda Avenue Grind and Overlay Project	\$75,537	
Description of Work		
CM Services		

The Local Agency of City of Fircrest

desires to supplement the agreement entered in to with <u>KPG, P.S.</u>

and executed on March 13, 2019 and identified as Agreement No. 19007

All provisions in the basic agreement remain in effect except as expressly modified by this supplement. The changes to the agreement are described as follows:

Section 1, SCOPE OF WORK, is hereby changed to read:

Provide CM services as detailed in Exhibit A, attached.

||

Section IV, TIME FOR BEGINNING AND COMPLETION, is amended to change the number of calendar days for completion of the work to read: <u>No Change - December 31, 2020</u>

Ш

Section V, PAYMENT, shall be amended as follows: Original Contract = \$37,750 + \$37,787 = \$75,537 (New contract amount)

as set forth in the attached Exhibit A, and by this reference made a part of this supplement. If you concur with this supplement and agree to the changes as stated above, please sign in the Appropriate spaces below and return to this office for final action.

By: Terry Wright	By:
/ inglight consultant Signature	Approving Authority Signature

EXHIBIT A-1

Construction Management Services Scope of Work - Supplement 1 July 1, 2019 City of Fircrest - ALAMEDA AVENUE GRIND & OVERLAY PROJECT Federal Aid Number: STPUL-2991 (003)

This work will provide construction management services for the construction contract to complete the City of Fircrest Alameda Ave Grind & Overlay project (hereinafter called "Project"). These services will include design support, project management, documentation control, inspection, materials testing, and contract administration during the construction of the project, as detailed below. KPG ("Consultant") will provide to the City of Fircrest ("City") construction management services for the project.

A detailed scope for the Contract follows:

I. INTRODUCTION

The following scope of services and associated costs are based upon the assumptions outlined below.

General Assumptions:

- The **proposed project team** will include a part-time project manager, one part-time documentation control specialist, a part-time project engineer during construction activities, one full-time inspector during construction activities when unit price work is being performed (which is anticipated to be 5 working days) and part time-time inspection during the remaining 10 working days.
- The level of service is based on project duration of approximately 15 working days of which 5 working days will require full time on-site inspection (9hrs/Day) and the remainder will only require part time inspection (3hrs/day).
- This scope of services provides for the management of the ROM and scheduling of required materials testing respectively.
- It is anticipated the Consultant will review all required insurance and bonding information for the contractor, check references and provide a recommendation of Award to the City.
- It is anticipated that all Community Outreach will be completed by the City.

EXHIBIT A-1

- The design engineers from KPG will be available during construction to answer questions during construction and review RAM's, shop drawings, and answer RFI's that require a level of engineering expertise outside of the capabilities of the City.
- Services will be performed in accordance with the contract plans & special provisions, WSDOT Local Agency Guidelines (LAG), the WSDOT Construction Manual, and City of Fircrest engineering standards.
- City/Field Office: No construction field office will be provided; it is anticipated that all construction meetings will be held at City offices.

II. SCOPE OF WORK

The objective and purpose of this Construction Management Services Agreement is for the Consultant to successfully deliver the construction of the Project to the City by ensuring that the improvements are constructed in accordance with the approved Plans and Specifications, as may be amended or revised, that all of the required Project documentation is accounted for, and ultimately that the City receives a successful review by WSDOT Local Programs at the end of the Project.

TASK 1 – MANAGEMENT/COORDINATION/ADMINISTRATION

Provide overall project management, coordination with the City, monthly progress reports, and invoicing. This effort will include the following elements:

- 1.1 Organize and layout work for project staff. Prepare project instructions on contract administration procedures to be used during construction.
- 1.2 Review monthly expenditures and CM team scope activities. Prepare and submit project progress letters to the City to accompany invoices that describe CM services provided each month. Prepare and submit reporting required by funding source(s), if any.

Deliverables

• Monthly invoices and progress reports

TASK 2 – PRECONSTRUCTION SERVICES

2.1 Preconstruction Conference: The Consultant will prepare an agenda for, distribute notices of, and conduct a preconstruction conference in the City's offices. The Consultant's project manager, resident engineer, inspector, and document control specialist will attend the preconstruction conference. The Consultant will prepare a written record of the meeting and distribute copies of the minutes to all attendees and affected agencies, staff, etc.

At the Pre-construction conference, the Consultant shall facilitate discussions with the Contractor concerning the plans, specifications, schedules, issues with utilities, unusual conditions, Federal, State, and local requirements, EEO, DBE and training requirements, and any other items that will result in better project understanding among the parties involved.

Deliverables

• Preconstruction conference agenda with meeting minutes

TASK 3 – CONSTRUCTION SERVICES - FIELD

- 3.1 The Consultant shall provide the services of part-time resident engineer (16 hours/week) during construction activities, one full time inspector during construction activities, and other duties necessary to monitor the progress of the work. Construction staff shall oversee work on the project site and will observe the technical progress of the construction, including providing day-to-day contact with the Contractor and the City. The following are major items of work anticipated:
 - Erosion Control
 - Pavement Repair
 - HMA overlay
 - Coordination of Temporary Traffic control
 - And all incidental items necessary to complete the Work as described in the Plans and/or Specifications.

Field inspection staff will perform the following duties as a matter of their daily activities:

- a) Observe technical conduct of the construction, including providing day-to-day contact with construction contractor, City, utilities, and other stakeholders, and monitor for adherence to the Contract Documents. The Consultant's personnel will act in accordance with Sections 1-05.1 and 1-05.2 of the Standard Specifications.
- b) Observe material, workmanship, and construction areas for compliance with the Contract Documents and applicable codes, and notify construction contractor of noncompliance. Advise the City of any non-conforming work observed while on site.
- c) Document all material delivered to the job site in accordance with the WSDOT LAG Manual.
- d) Prepare daily inspection reports, recording the construction contractor's operations as actually observed by the Consultant; includes quantities of work placed that day, contractor's equipment and crews, and other pertinent information. All daily inspection reports will adhere to WSDOT Local Agency Guidelines.
- e) Interpret Contract Documents in coordination with City and KPG.
- f) Resolve questions which may arise as to the quality and acceptability of material furnished, work performed, and rate of progress of work performed by the construction contractor.
- g) Establish communications with adjacent property owners. Respond to questions from property owners and the general public.

- h) Coordinate with permit holders and insure compliance with approved permits, if applicable.
- i) Prepare field records and documents to help assure the Project is administered in accordance with funding agency requirements. The utilities will provide measurement and payment information to the consultant for development of monthly pay estimates.
- j) Collect and calculate delivery tickets and salesman's daily reports of aggregate. All tickets will be initialed, and include the correct bid item, as well as identify the stationing (Construction Manual 10-2).
- k) Attend and actively participate in regular on-site weekly construction meetings.
- 1) Take periodic digital photographs during the course of construction, and record locations.
- m) Coordinate with the City's traffic operations and maintenance personnel.
- n) Conduct wage rate interviews, DBE interviews and training interviews in accordance with the LAG Manual.
- O) Upon substantial completion of work, coordinate with the Client and affected agencies to prepare a 'punch list' of items to be completed or corrected. Coordinate final inspection with those agencies.
- p) Provide oversight during audits performed by WSDOT Local Programs and others.

Assumptions:

- Consultant will provide observation services for the days/hours that the contractor's personnel are on-site.
- The Consultant's monitoring of the construction contractor's activities is to ascertain whether or not they are performing the work in accordance with the Contract Documents; in case of noncompliance, Consultant will reject non-conforming work, and pursue the other remedies in the interests of the City, as detailed in the Contract Documents. The Consultant cannot guarantee the construction contractors' performance, and it is understood that Consultant shall assume no responsibility for: proper construction means, methods, techniques; project site safety, safety precautions or programs; or for the failure of any other entity to perform its work in accordance with laws, contracts, regulations, or City's expectations.
- 3.2 **Materials Testing Coordination:** Coordinate the work of the materials testing technicians and testing laboratories in the observation and testing of materials used in the construction; document and evaluate results of testing; and address deficiencies. Frequency of testing shall be as required in the WSDOT Construction Manual and LAG manual.

Qualified tester requirements as identified in WSDOT LAG 52.3.32 illustrate that for projects on non-NHS highway systems, there is no requirement for qualified testers, but the requirements within the manual must be followed if federal funding applies.

Assumptions

It is assumed that no material fabrication inspection per WSDOT LAG 52.3.32 will be required.

EXHIBIT A-1

3.2 **Substantial Completion:** Upon substantial completion of work, coordinate with the City and other affected agencies, to perform a project inspection and develop a comprehensive list of deficiencies or punchlist of items to be completed. A punchlist and Certificate of Substantial Completion will be prepared by the Consultant and issued by the City.

Deliverables

- Daily Construction Reports with project photos submitted on a weekly basis
- Punch List, Certificate of Substantial Completion

TASK 4 – CONSTRUCTION SERVICES – OFFICE

4.1 **Document Control.** Original documentation will be housed at the Consultant's office, and filed in accordance with standard filing protocol to meet WSDOT Highways & Local Program Requirements. A copy of working files will be maintained in the field office.

Document Control also consists of preparing Final Project Reports for the City for WSDOT and FHWA acceptance and include:

- Final Estimate (Approving Authority File)
- Comparison of Preliminary and Final Quantities (Approving Authority File)
- Final Records as identified in WSDOT LAG (Approving Authority File)
- Record of Material Samples and Tests
- Materials Certification
- Affidavit of Wages Paid
- Release for the Protection of Property Owners and General Contractor
- DOT Form 422-102 EF, Quarterly Report of Amounts Credited as DBE Participation
- Final Estimate (Approving Authority File)
- Comparison of Preliminary and Final Quantities (Approving Authority File)
- Final Records as identified in WSDOT LAG (Approving Authority File)
- Record of Material Samples and Tests
- Materials Certification
- Affidavit of Wages Paid
- Release for the Protection of Property Owners and General Contractor
- DOT Form 422-102 EF, Quarterly Report of Amounts Credited as DBE Participation
- 4.2 **Project Coordination**: Liaison with City, construction contractor, engineer, utilities and property owners on a regular basis to discuss project issues and status.
- 4.3 **Plan Interpretations:** Provide technical interpretations of the drawings, specifications, and Contract Documents, and evaluate requested deviations from the approved design or specifications. Coordinate with City for resolution of issues involving scope, schedule, and/or budget changes.
- 4.4 Weekly Meetings: Lead weekly meetings, including preparation of agenda, meeting minutes, and distribution of minutes to attendees. Outstanding issues will be tracked on a weekly basis. (only two meetings are assumed)

- 4.5 **Initial Schedule Review**: Perform detailed schedule review of The Contractor provided CPM for conformance with the Contract Documents.
- 4.6 Lump Sum Breakdown: Will not be required due to the short project duration.
- 4.7 Monthly Pay Requests: Prepare monthly requests for payment, review with the City, Contractor, and approve as permitted. Utilize City provided format for pay estimates, or Consultant format. (only one Pay Estimated is assumed)
- 4.8 Monthly Schedule Review: At the monthly cutoff, review contractor's updated schedule and compare with field-observed progress, as described in Section 1-08 of the Special Provisions. In addition, perform schedule analysis on contractor provided CPM updates and review schedule for delays and impacts. Coordinate with Contractor in the development of recovery schedules, as needed, to address delays caused by either events or issues within the Contractor's control or other events or issues beyond the Contractor's control.
- 4.9 **Certified Payroll**: Process and track all certified payroll per WSDOT Highways & Local Program Requirements. This includes verifying the initial payroll for compliance and 10% of all payrolls submitted thereafter and tracking payroll each week. Payroll and payroll logs will adhere to WSDOT LAG requirements.
- 4.10 Weekly Statement of Working Days: Prepare and issue weekly statement of working day report each week.
- 4.11 WSDOT & FHWA Reporting: Prepare and track all necessary reports per WSDOT Highways & Local Program Requirements. This will include DBE condition of award tracking throughout the project, DBE Reporting, EEO reporting, training Goal tracking via certified payroll, DBE on site reviews, Training Questionnaires, and periodically meeting with WSDOT Highways and Local Programs staff as requested to review project status.
- 4.12 **Subcontractor Documentation**: Process / Approve all required subcontractor documentation per WSDOT Highways & Local Program Requirements. Request to Sublets will be verified and logged. Certification for Federal Aid will be obtained from all subcontractors. This includes checking System Award Management System (SAMS), verifying OM/WBE, verifying business licensing, reviewing insurance documentation, verifying city business licensing, Intent to Pay Prevailing Wage and Affidavit of Wages Paid. All subcontractor documentation will be logged into WSDOT's subcontractor logs.
- 4.13 **Cost Projections:** Will not be required due to the short project duration.
- 4.14 **Record Drawings:** It is assumed Record Drawings will not be required.
- 4.15 Audit Oversight: Provide oversight during audits performed by WSDOT Local Programs and others.

- 4.16 **Physical Completion Letter:** Following completion of all punchlist work, prepare physical completion letter to the contractor and WSDOT, and recommend that City and/or utilities accept the project.
- 4.17 **Project Closeout:** Transfer all project documents to the City for permanent storage.
 - Schedule review comments
 - Meeting agendas and notes
 - Monthly Pay Estimates
 - WSDOT Reporting
 - Subcontractor Packets
 - Physical Completion Letter
 - Final Project Documents

TASK 5 – SUBMITTAL/RFI PROCESSING

- 5.1 **Submittals:** Coordinate review process for shop drawings, samples, traffic control plans, test reports, and other submittals from the Contractor for compliance with the Contract Documents. Route key submittals to the City for their review and approval. Submittals shall be logged and tracked.
- 5.2 **Request for Information (RFI):** Review and respond to RFI's. RFI's shall be logged and tracked.
- 5.3 **Record of Materials (ROM):** <u>Utilize ROM prepared by KPG</u> and update based on special provisions and plans for use on the project, based on the contract specifications and WSDOT/LAG requirements. Maintain the ROM according to WSDOT Highways & Local Program Requirements. The ROM will track all of the materials delivered to the site including manufacturer/supplier, approved RAM's, QPL items, material compliance documentation, and all other required documentation.

Deliverables

- Submittal log
- RFI Log
- Completed Record of Material for Material Certification and updated as needed

TASK 6 – CHANGE MANAGEMENT

- 6.1 **Case Log:** Develop and maintain a case log which includes change orders, RFP's, Field Work Directives per Highway & Local Programs Guidelines
- 6.2 **Change Orders:** Develop change orders per Highway & Local Programs Guidelines, provide technical assistance to negotiate change orders, and assist in resolution of disputes which may occur during the course of the project. Each change order will be executed in accordance with WSDOT Local Agency Guidelines and contain the following:
 - Change order
 - Independent Cost Estimate
 - Time Impact Analysis
 - Contractor's Pricing
 - Verbal Approval Memo

- Change Order Checklist (LAG Manual)
- Change Order Request Form
- Back up documentation
- 6.3 **Field Work Directives:** Prepare field work directives as necessary in order to keep the contractor on schedule.
- 6.4 **Minor Change Orders:** Develop minor change orders per WSDOT Highways & Local Program Requirements. Each minor change order will be executed in accordance with WSDOT Local Agency Guidelines and contain the following:
 - Independent Cost Estimate
 - Verbal Approval Memo
 - Back up documentation
- 6.5 **Force Account:** Track contractor force account labor, equipment and materials and provide payment according to WSDOT requirements. All force account calculations will be verified by the engineer and double checked by the documentation specialist per WSDOT LAG Manual.

Deliverables

- Change Order(s)
- Case Management Log
- RFI Log
- Minor Change Order(s)
- Force Account Records

FEE SUMMARY

Supplement #1

Project: City of Fircrest Alameda Ave Grind & Overlay Construction Services Fee - (7/1/19) STPUL-3118(001)

Description		Estimated Fee		
Task 1 - Management/Coordination/Administration		\$	2,132.00	
Task 2 - Preconstruction Services		\$	2,602.00	
Task 3 - Construction Services - Field		\$	12,640.00	
Task 4 - Construction Services - Office		\$	13,595.00	
Task 5 - Submittal/RFI Processing		\$	3,918.00	
Task 6 - Change Management		\$	2,900.00	
	Subtotal:	\$	37,787.00	
Task 7 - Management Reserve	MR:	\$		
	Total:	\$	37,787.00	

HOUR AND FEE ESTIMATE

	City of Fircrest Alameda Ave Grind & Overlay Construction Services Fee - (7 STPUL-2991(003)	Landscape Architecture							
	15 = working days			Labor Hou	r Estimate	9		To	otal Fee
Task	3 Weeks Description	Principal	Resident Engineer	*Const Observer II	*Doc Specialist	*Const Office Eng	*Const Assistant		
		\$ 232.00	\$ 125.00	\$ 110.00	\$ 115.00		\$ 77.00		Fee
Toek	1 - Management/Coordination/Administ	Instian							
1.1	Project Organization and Layout	2		1	2		8	\$	1,310
1.2	Monthly Review and Invoicing (2 month	2			2		4	\$	772
1.2	Reimbursable expenses - see breakdow		le				4	Ф \$	50
	Task Totals	4	0	0	2	0	12	S	2,13
				how we want				_	
	2 - Preconstruction Services			_					
2.1	Preconstruction Conference	1	4	4	12			\$	2,552
								\$	
	Reimbursable expenses - see breakdow							\$	50
_	Task Total	1	4	4	12	0	0	\$	2,602
Task	3 - Construction Services - Field								
3.1	Construction Observation			75		-		S	8,250
3.2	Material Testing		6	10	2			\$	980
3.3	Substantial Completion		4	2	6				
0.0		at a star	4	2	D)	\$	1,410
-	Reimbursable expenses - (Materials Tes Task Total	aing) D	10	77		-	0	\$	2.000
	Task Total	U	10	77	8	0	U	\$	12,640
Task -	4 - Construction Services - Office								
4.1	Document Control	_			24	1	12	\$	3.684
4.2	Project Coordination	1			2	()	12	\$	462
4.3	Plan Interpretation		2		-			\$	250
4.4	Weekly Meetings (2)	2			0		0		
4.5	Initial Schedule Review	4	4		8		6	\$	2,346
_			1		1			\$	240
4.6	Lump Sum Breakdown						_	\$	
4.7	Monthly Pay Requests (est. 1)		2		6		2	\$	1,094
4.8	Monthly Schedule Review							\$	•
4.9	Certified Payroll				6		4	\$	998
4.10	Weekly Statement of Working Days				3			\$	345
4.11	WSDOT & FHWA Reporting				4	1		\$	460
4,12	Subcontractor Documentation			1	6		2	\$	844
4,13	Cost Projections							\$	
4.14	Check Record Drawings Progress							\$	
4,15	Audit-Oversight							\$	
4.16		1	2		2			\$	712
4.17	Project Closeout	1	4		8		4	\$	1,960
	Reimbursable expenses - see breakdow	n for detai	ls				· · · · · · · · · · · · · · · · · · ·	ŝ	200
	Task Total	5	15	0	70	0	30	\$	13,595
Task	- Submittal/RFI Processing								
5.1	Submittals		4		8			\$	1,420
5,2	Request for Information (RFI)	1	2		2			\$	712
5.3	Record of Materials (ROM)		2		8		8	\$	1,786
	Reimbursable expenses - see breakdow	n for detai	ls					\$	
	Task Total	î	8	0	18	0	8	\$	3,918
Task 6	5 - Change Management							_	
	Case Log			1				\$	
6.2	Change Orders		8		4			\$	1,460
6.3	Field Work Directives		-					\$.,
6.4	Minor Change Orders		4		2			\$	730
6.5	Force Account		2		4			\$	710
	Reimbursable expenses - see breakdow	n for data	_					\$	
	nominuoisavio expenses - sec preakting								

Total: \$ 37,787

HOUR AND FEE ESTIMATE



Project:	City of Fircrest			
	Alameda Ave Grind & Overlay			
	Construction Services Fee - (7/1/19)			
	STPUL-2991(003)			

Reimbursable Breakdown		Cost
Task 1 - Management/Coordination/Administration		
Mileage	\$	50.00
Reproduction		
Task 1 - Total	\$	50.00
Task 2 - Preconstruction Services		
Mileage	\$	50.00
Reproduction		
Task 2 - Total	\$	50.00
Task 3 - Construction Services - Field		
Construction Vehicle (@ \$750/mo)	\$	-
Misc. Field Supplies	\$	
Materials Testing	\$	2,000.00
	\$	
Task 3 - Total	\$	2,000.00
Task 4 - Construction Services - Office		
Mileage	\$	200.00
Misc Office Supplies	\$	
Task 4 - Total	\$	200.00
Task 5 - Submittal/RFI Processing		
Mileage	\$	-
Reproduction	\$	
Task 5 - Total	\$	-
Task 6 - Change Management		
Mileage	\$	-
Reproduction	\$	-
Task 6 - Total	S	

Total Reimbursable Costs: \$ 2,300.00